

University of Huddersfield Repository

Saad, Syed Mohammad

Experimental and Numerical Investigations on the Transient Performance of a Turbocharged Diesel Engine

Original Citation

Saad, Syed Mohammad (2020) Experimental and Numerical Investigations on the Transient Performance of a Turbocharged Diesel Engine. Doctoral thesis, University of Huddersfield.

This version is available at http://eprints.hud.ac.uk/id/eprint/35423/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

EXPERIMENTAL AND NUMERICAL INVESTIGATIONS ON THE TRANSIENT PERFORMANCE OF A TURBOCHARGED DIESEL ENGINE

Syed Mohammad Saad

School of Computing and Engineering University of Huddersfield

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of

Doctor of Philosophy

October 2020

Copyright

i. The author of this thesis (including any appendices and/or schedules to this thesis) owns any copyright in it (the "Copyright") and he has given The University of Huddersfield the right to use such Copyright for any administrative, promotional, educational and/or teaching purposes.

ii. Copies of this thesis, either in full or in extracts, may be made only in accordance with the regulations of the University Library. Details of these regulations may be obtained from the Librarian. This page must form part of any such copies made.

iii. The ownership of any patents, designs, trade marks and any and all other intellectual property rights except for the Copyright (the "Intellectual Property Rights") and any reproductions of copyright works, for example graphs and tables ("Reproduction"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property Rights and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property Rights and/or Reproductions.

Abstract

In the highly competitive automotive market the share of diesel engine is continuously increasing. They are extensively used in passenger cars as well as in long distance haulage sector vehicles to impart motive force. Regulations for diesel engine emissions as well as public concern for fuel economy have forced the research community to address the combustion and emission issues associated with the use of the diesel engines in general and during the transient operation mode in particular. Turbocharging of diesel engine is the most extensively used technology for the improvement of power density, emissions and enabling downsizing of engines without compromising power. However, turbocharged diesel engine suffer from turbo-lag which is a common phenomenon especially during the rapid transient conditions. Turbo lag causes engine performance deterioration and increased emissions during transient events. The studies on reducing the turbo lag and maintaining the desirable air fuel ratio are considered to be very important for making turbocharged engine compliant with the current environmental regulations.

The thesis covers the important aspects of a turbocharged diesel engine with a specific focus on the transient response of the diesel engine. In the first facet of this research, computer-based investigations using commercial engine simulation packages are performed to simulate the transient response of the system using different methods. Torque assistance of 0.16 Nm reduces the turbo lag by 3.6 sec for both compressor exit pressure and compressor speed. Optimum value of inertia reduction is found to be -10% which reduces the turbo lag by 2.9 sec for compressor exit pressure and by 0.6 sec for compressor speed. The effect of 2.5 bar air injection is found to reduce the turbo lag by 3.9 sec for both compressor exit pressure and compressor speed. A comparison is made for the assessment of relative improvement in transient response brought by the three methods and based on this comparison, air injection system is chosen that yields maximum benefit in the performance of the engine. The experimental results from the real CI engine is used to validate the simulation model. A good agreement is achieved between the simulation and experimental results.

The effect of air injection on a heavy-duty turbocharged diesel engine under various operating conditions forms the second facet of this study. The turbocharger response parameters are analyzed under the effect of air injection for different transient operating conditions of speed and load transients. For speed transients, considering the energy imparted for air injection, 1 bar is the optimum injection pressure with turbo lag reduction per unit energy as 0.290, 0.392 and 0.555 per joule for compressor exit pressure, turbine inlet pressure and turbine inlet temperature (TIT) respectively. Faster recovery time is noted for 1 sec rapid acceleration than that of 2 sec with the application of air injection. For load transients, optimization of air injection is also performed for injection pressure and orifice diameter. Under constant load step, maximum improvement in turbo lag reduction is observed at 1000 -1400 rpm whereas its effect on maximum attainable value is at 1600 - 1800rpm. For load magnitude variations, the optimum injection pressure of 3 bar at 15 mm orifice diameter brings more improvement for lighter load than for stronger load. Maximum improvement is noted for 50-70% load step. 3 bar air injection at 15mm orifice diameter is the optimum injection pressure which brings more improvement in terms of turbo lag reduction for faster load application. The effect of this optimized air injection is more beneficial for 1 sec load schedule than for 2 sec.

During the transient operation of diesel engines, exhaust emissions are the imperative issue that needs to be addressed. In the third facet of the research exhaust emissions are analyzed to make sure that the application of air injection technique does not need compromise on emissions. The air injection technique is evaluated for exhaust emissions through simulation. 3 bar air injection at 10 mm orifice diameter is the optimum air injection for emissions under speed transient which satisfies Euro 6 standard for CO and HC emissions. This air injection reduces the concentration of CO and HC emissions by 5% and 0.4% respectively. Under load transient the optimum value of air injection is 1.2 bar at 10 mm orifice diameter which reduces the concentration of CO, HC and NO_x emissions by 0.8%, 0.01% and 0.4% respectively.

The study reveals that air injection technique while improving the transient response of the turbocharged diesel engine doesn't increase the emissions, highlighting the magnitude of contribution of the technique to the overall performance of the system. The novelty in the research is the compilation of these methods in a cohesive approach of modeling the transient response of turbocharged engine system.

Acknowledgements

First and above all, I would like to thank Allah Almighty for giving me the strength, knowledge and opportunity to undertake this research and most importantly to persevere despite the pressures on my time. I take pride in expressing gratitude to my parents who always visionalised my success and their dream for my degree propelled me to proceed successfully.

I want to express my heartiest gratitude and sincere thanks to my supervisor Professor Rakesh Mishra for his continuous supervision, insightful guidance and overall direction. I am deeply indebted to him for the positive feedbacks, timely guidance and useful advises. His detailed review for the correction of my manuscripts has certainly helped me in writing this thesis. He always provided directions to prevent the research going in diverted ways. This thesis would not have been possible without his valuable guidance and support. My sincere gratitude to Dr Taimoor Aasim for his useful suggestions and helpful discussions on my work.

It would be inappropriate if I don't acknowledge the help of Ricardo Support Team for providing me useful solutions to the technical errors I faced during my simulation work on Ricardo wave. I would also like to convey special thanks to Lotus Engine Simulation team for providing technical support during the initial phase of my research. My thanks also goes to Mr Martin Gargett for assisting me in the data collection phase on the engine test rig at the Advanced Automotive Laboratory of University of Huddersfield. I like to extend my thanks to School of Computing and Engineering technicians and IT Support of the University of Huddersfield for resolving issues related to remote desktop connections.

My acknowledgement would be incomplete without thanking my better half Dr Asiya Rummana whose continuous encouragement during the challenging moments of this journey was the driving force for this achievement. I would like to thanks my two beautiful kids - my daughter Fatima Saad and my son Syed Misbah Ul Hasan for their patience and enormous corporation during this research.

Further, I would like to acknowledge the support from Ibra College of Technology for providing a positive and encouraging work environment to conduct the research. Last but not the least, I would like to thank everyone who directly or indirectly prayed for me along the way. Dedicated to my late parents - Mr Sayeed Ul Hasan and Mrs Noorus Sabah

Declaration

I declare that I have presented my own work in this thesis and relevant citations have been included to acknowledge the work of others. I have two papers published in the journals and I have referred them wherever the material from these papers are reproduced in the thesis.

Publications and contributions

 Saad, S. M. and Mishra, R. (2019). Performance of a Heavy-Duty Turbocharged Diesel Engine Under the Effect of Air Injection at Intake Manifold During Transient Operations. Arabian Journal for Science and Engineering.

I developed the engine model, performed data analysis, interpreted the results and presented in manuscript form.

 Mishra, R., and Saad, S. M. (2017). Simulation based study on improving the transient response quality of turbocharged diesel engines. Journal of Quality in Maintenance Engineering. ISSN 1355-2511.

I developed the engine model, validated the steady state results against the experimental data which I collected from the test bed at University of Huddersfiled, developed the test strategies and wrote the manuscript.

Contents

Li	List of Figures 1			
Li	st of	Tables	17	
1	INT	TRODUCTION	19	
	1.1	Turbocharged diesel engine	21	
		1.1.1 Geometry and performance of turbocharged diesel engines	23	
	1.2	Transient response of turbocharged diesel engine	24	
	1.3	Diesel engine exhaust emissions during transient operation	26	
	1.4	Research problem and aims	27	
	1.5	Thesis structure	29	
2	BA	CKGROUND AND LITERATURE REVIEW	31	
	2.1	Transient operating conditions	31	
	2.2	Existing methods for controlling turbo lag	34	
		2.2.1 Electric torque assistance	39	
		2.2.2 Inertia reduction of the turbocharger $\ldots \ldots \ldots \ldots \ldots \ldots$	40	
		2.2.3 Air injection strategies	40	
	2.3	Diesel engine exhaust emissions under transient events	43	
	2.4	Summary and knowledge gap on methods of controlling turbo lag	46	
	2.5	Aims and objectives of the study	47	
3	EN	GINE SIMULATION	49	
	3.1	Lotus engine simulation	51	
	3.2	Ricardo wave simulation	54	
	3.3	Modelling of techniques	57	
		3.3.1 Modelling of torque assist and inertia reduction	57	

CONTENTS

		3.3.2	Modelling of air injection	. 58
	3.4	CI eng	gine test bed	. 60
	3.5	Test p	procedures	. 62
		3.5.1	Strategies for transient events	. 62
			3.5.1.1 Fueling rate change \ldots \ldots \ldots \ldots \ldots \ldots	. 62
			3.5.1.2 Speed transients \ldots \ldots \ldots \ldots \ldots \ldots \ldots	. 63
			3.5.1.3 Load transients \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	. 63
			3.5.1.3.1 Constant load \ldots \ldots \ldots \ldots \ldots \ldots	. 64
			$3.5.1.3.2$ Load magnitude \ldots \ldots \ldots \ldots \ldots	. 65
			$3.5.1.3.3$ Load schedule \ldots \ldots \ldots \ldots \ldots	. 67
	3.6	Measu	rement uncertainty	. 68
	3.7	Summ	hary of the chapter \ldots	. 69
4	БТТ		IC TODOLLE ACCIETANCE AND DEDUCTION IN IN	
4			E COMPRESSOR WHEEL	- 71
	ER .	Pocult	ts and discussion	72
	4.1	A 1 1	Steady state simulations	. (ə 72
		4.1.1	Transient simulation	. 75
	12	4.1.2 Summ	mansient simulation	. 70
	4.2	wheel	ary on electric torque assist and reduction in mertia of compresso	80
		wneer		. 02
5	Air	inject	ion strategies	85
	5.1	Detail	led investigation on air injection systems	. 86
		5.1.1	Validation with real engine	. 86
	5.2	Air in	jection system and its comparison with other methods of turbo lag	S
		reduct	tion \ldots	. 90
	5.3	Evalua	ation of air injection system under acceleration test $\ldots \ldots$. 94
		5.3.1	Test 1 \ldots	. 95
		5.3.2	Test 2	. 98
		5.3.3	Test 3	. 103
	5.4	Evalua	ation of air injection system under load test	. 105
		5.4.1	Turbocharger response for constant load step $\ldots \ldots \ldots$. 106
		5.4.2	Load magnitude variations	. 111
			5.4.2.1 Effect of air injection for varying load magnitude	. 111

			5.4.2.2	Optimization of injection pressure $\ . \ . \ . \ . \ .$.	113
			5.4.2.3	Effect of air injection on turbocharger response param-	
				eters	118
		5.4.3	Load sch	edule	119
	5.5	Summ	ary of the	e chapter	120
6	EFI	FECT	OF AIR	INJECTION ON EMISSION CHARACTERIS-	
	TIC	SOF	A CI EN	IGINE DURING TRANSIENT OPERATION	123
	6.1	Emissi	on charac	teristics under speed transient	124
		6.1.1	Optimiza	ation of air injection for emissions under speed transient	124
		6.1.2	Emission	reduction under the effect of air injection for speed tran-	
			sient		127
	6.2	Emissi	on charac	teristics under load transient $\ldots \ldots \ldots \ldots \ldots \ldots$	128
		6.2.1	Optimiza	ation of air injection for emissions under load transient .	128
		6.2.2	Emission	reduction under the effect of air injection for load transien	t131
	6.3	Overal	ll optimiza	ation of air injection system	132
	6.4	Summ	ary of the	e chapter	133
7	CO	NCLU	SIONS		135
	7.1	Major	conclusio	ns and thesis contribution	135
		7.1.1	Develop	nent of a novel strategy for turbo lag reduction \ldots .	135
		7.1.2	Effect of	air injection on the transient response of diesel engine	
			under va	rious operating conditions	136
		7.1.3	Effect of	air injection on environmental performance of the engine	e 137
	7.2	Limita	tions and	scope for future work	138
Re	efere	nces			139

List of Figures

1.1	P-V and T-S diagram of four stroke CI engine	20
1.2	Comparison of brake torque in turbocharged and naturally aspirated	
	engine [21]	22
1.3	Comparison of brake power in turbocharged and naturally aspirated en-	
	gine [21]	22
1.4	Turbocharged diesel engine [24]	23
1.5	Layout of turbocharged diesel engine	24
1.6	Comparison between steady and transient operation of diesel engine [16]	25
2.1	Effect of load change on final speed and recovery period $[47]$	33
2.2	Effect of load schedule on final speed and recovery period $[47]$	33
2.3	Combined supercharging $[16]$	35
2.4	Two-stage turbocharging $[16]$	36
2.5	Variable geometry turbine (VGT) [16]	36
2.6	Sequential turbocharging $[16]$	38
2.7	Electric torque assist (ETA) [16] $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	38
2.8	NO concentration in the exhaust, maximum gas temperature and oxygen	
	concentration in the fuel spray $[40]$	44
3.1	Steps in simulation modelling	50
3.2	Turbocharged engine model in Lotus Engine	53
3.3	Turbocharged engine model in Ricardo Wave	56
3.4	Wave coupled Simulink Model	60
3.5	Schematic of experimental Facility [98]	61
3.6	Identification of load steps in load tests	65
3.7	Variation of engine torque with different load steps in 1 sec, initial load	
	10%	66
3.8	Variation of brake torque with different load steps in 1 sec, initial load	
	50%	66

LIST OF FIGURES

3.9	Variation of brake torque with different load steps in 2 sec, initial load 10%	67
3.10	Variation of brake torque with different load steps in 2 sec, initial load 5007	60
	50%	08
4.1	Block diagram for the proposed system	72
4.2	Brake torque	74
4.3	Brake mean effective pressure	74
4.4	Brake specific fuel consumption	75
4.5	Brake power	75
4.6	Compressor exit pressure response for different value of torque assist $\ . \ .$	77
4.7	Compressor exit pressure with and without torque assist $\ldots \ldots \ldots$	78
4.8	Rate of change of Compressor outlet pressure response	78
4.9	Compressor speed response with torque assist and without torque assist	79
4.10	Rate of change of compressor speed response	79
4.11	Compressor exit pressure with reduction in inertia values $\ldots \ldots \ldots$	81
4.12	Compressor Speed with reduction in inertia values	81
5.1	Strategies to model the transient response of diesel engine	86
5.2	Validation of simulated values of brake torque with experimental data $% f(x)=\int dx dx$.	87
5.3	Validation of simulated values of brake specific fuel consumption with	
	experimental data	88
5.4	Validation of simulated values of brake power with experimental data $\ .$	89
5.5	Variation of brake torque with injection pressure	90
5.6	Variation of brake power with injection pressure	91
5.7	Effect of air injection pressures on compressor exit pressure	91
5.8	Effect of air injection pressures on compressor speed $\ldots \ldots \ldots \ldots$	92
5.9	Comparison of compressor exit pressure response (top) and compressor	
	speed response (bottom) with torque assist method, air injection method,	
	reduction in inertia method and original (no method applied) \ldots .	93
5.10	Effect of acceleration time on compressor exit pressure (CEP) response .	96
5.11	Effect of acceleration time on turbine inlet pressure (TIP) response	96
5.12	Effect of acceleration time on turbine inlet temperature (TIT) response	97
5.13	Variation of response parameters (CEP, TIP and TIT) with acceleration	
	time	97
5.14	Effect of air injection pressures on compressor exit pressure response	98
5.15	Effect of air injection pressures on turbine inlet pressure response	99

LIST OF FIGURES

5.16	Effect of air injection pressures on turbine inlet temperature response .	99
5.17	Variation of response parameters with air injection pressures for turbo lag1	00
5.18	Dependency of turbo lag reduction on air supply parameters for com-	
	pressor exit pressure (CEP)	01
5.19	Dependency of turbo lag reduction on air supply parameters for turbine	
	inlet pressure (TIP) 1	02
5.20	Dependency of turbo lag reduction on air supply parameters for turbine	
	inlet temperature (TIT)	02
5.22	Rate of change of compressor exit pressure	.03
5.21	Effect of air injection on response parameters with variable acceleration	
	time	04
5.23	Rate of change of turbine inlet pressure	04
5.24	Rate of change of turbine inlet temperature	05
5.25	Schematic of load tests	06
5.26	Effect of air injection on compressor exit pressure response during a load	
	change from 80% to 100% at different engine speeds \hdots	.07
5.27	Effect of air injection on turbine inlet pressure response during a load	
	change 80% to 100% at different engine speeds $\ldots \ldots \ldots \ldots \ldots \ldots \ldots $	07
5.28	Effect of air injection on turbine inlet temperature response during a	
	load change 80% to 100% at different engine speeds $\ldots \ldots \ldots \ldots 1$	08
5.29	Dependency of turbo lag and SS value ratio on engine speed for com-	
	pressor exit pressure (CEP) 1	09
5.30	Dependency of turbo lag and SS value ratio on engine speed for turbine	
	inlet pressure (TIP) 1	09
5.31	Dependency of turbo lag and SS value ratio on engine speed for turbine	
	inlet temperature (TIT) 1	10
5.32	Dependency of transient response improvement on engine speed in terms	
	of turbo lag reduction (%) $\ldots \ldots 1$	10
5.33	Effect of air injection on engine speed with different load steps, initial	
	load 10%	12
5.34	Effect of air injection on engine speed with different load steps, initial	
	load 50%	12
5.35	Engine speed with varying orifice diameter and injection pressures for	
	load test L1	14
5.36	Correlation between turbo lag, injection pressure and orifice diameter	
	for L1 test	15

5.37	Correlation between turbo lag, injection pressure and orifice diameter	
	for L4 test	116
5.38	Correlation between turbo lag, injection pressure and orifice diameter	
	for L5 test	117
5.39	Correlation between turbo lag, injection pressure and orifice diameter	
	for L6 test	117
5.40	Effect of air injection on turbocharger response parameters for different	
	load magnitude	118
5.41	Engine speed with varying orifice diameter and injection pressures for	
	load step 50-70% in 2 sec \ldots	119
5.42	Effect of air injection on turbocharger response parameters for different	
	load schedules for L6 load test	120
6.1	CO emissions with varying injection pressures and orifice diameter	125
6.2	HC emissions with varying injection pressures and orifice diameter	125
6.3	NO ₂ emissions with varying injection pressures and orifice diameter	126
6.4	Optimization of air injection for CO and HC emissions at 1700 rpm	-
	engine speed	127
6.5	Effect of air injection on emission concentration under speed transient:	
	CO emission (left), HC emission (right) and NO_x emission (bottom)	128
6.6	CO emissions with varying injection pressures and orifice diameter	129
6.7	HC emissions with varying injection pressures and orifice diameter	130
	The emissions with varying injection pressures and office diameter	100
6.8	NO_2 emissions with varying injection pressures and orifice diameter \ldots	130

List of Tables

2.1	Various methods of improving the transient response with their pros and	
	cons	42
3.1	Engine specification	54
3.2	Engine block specifications	61

LIST OF TABLES

3.3	Basic valve parameters
3.4	Test cases
3.5	Tests cases for load strategies
3.6	Load tests resulting in torque deficit
4.1	Quality factors for transient test
4.2	Comparison of the results obtained by torque assist method. TL: Turbo
	lag
4.3	Reduction in compressor inertia
4.4	Comparison of the results obtained by inertia reduction method. TL:
	Turbo lag
5.1	Comparison of Ricardo wave results with experimental data for perfor-
	mance parameters
5.2	Comparison of turbo lag observed for compressor exit pressure (CEP)
	and compressor speed (C_Speed) response by different methods $\ldots $ 94
5.3	Turbo lag reduction per unit energy corresponding to air injection pres-
	sure for turbocharger response parameters (CEP, TIP and TIT) 103
5.4	Dependency of injection pressure (IP in bar) and turbo lag (TL in sec)
	on orifice diameter(mm) for the different load steps 114
5.5	Turbo lag reduction $(\%)$ for the turbocharger response parameters under
	different load tests
6.1	Dependency of CO emissions on injection pressure and on orifice diameter 131
6.2	Dependency of HC emissions on injection pressure and on orifice diameter 131
6.3	Dependency of NO_2 emissions on injection pressure and on orifice diameter 131
6.4	Quantification of the effect of air injection in terms of energy imparted
	per second. P: injection pressure (bar), OD: Orifice diameter (mm) 132

1

INTRODUCTION

Invention of internal combustion engine is considered as one of the most significant inventions of the last century which had a significant impact on the society, particularly on human mobility. It became the foundation of the successful development of several commercial technologies. It has transformed the entire transportation industry with its relatively low cost, high efficiency and favorable power to weight ratio [1].

Diesel engines are also known as compression ignition (CI) engines. The air that is sucked from the intake manifold into the cylinder is compressed to above the auto ignition temperature of the fuel. At the end of the compression stroke the fuel is sprayed into the combustion chamber under high pressure and temperature conditions and combustion takes place. The energy hence released from the burning of fuel forces the piston which moves from top dead center to bottom dead center and this motion is converted to forward motion of the vehicle through integrated and well design power train. Compression ignition engines plays a vital role in transportation industry where a very high amount of torque is required such as trucks and buses etc. [2]. There are four processes involved in diesel cycle as shown in Figure 1.1:

- 1-2: Isentropic compression
- 2-3: Constant pressure heat addition
- 3-4: Isentropic expansion
- 4-1: Constant pressure heat rejection

Intake stroke: The intake valve opens as the piston moves from top dead center to bottom dead center and the air is introduced inside the cylinder.

Compression stroke: The movement of the piston back to top dead center from the bottom dead center compresses the air inside the cylinder and the temperature inside the cylinder is raised over the auto-ignition temperature of the fuel.

1. INTRODUCTION

Power stroke: The fuel is injected into the combustion chamber by the injectors at the end of the compression stroke. The fuel auto ignites after the ignition delay period. The mixture of fuel and air combust to produce pressure on the cylinder head which pushes the cylinder downward.

Exhaust stroke: After completion of the power stroke, the piston moves from bottom dead center to top dead center and the exhaust valve opens. Due to the sweep effect of the piston, the exhaust gas leaves the cylinder.



Figure 1.1: P-V and T-S diagram of four stroke CI engine

The speed and power of the diesel engines are regulated by the amount of fuel injected [3]. Due to their operational simplicity, diesel engines are more preferable in the transport sector, agriculture and forestry machinery [4].

The excellent fuel efficiency is the most attractive feature of the compression ignition engine which can surpass 40% in vehicle application and 50% in large two stroke units for marine propulsion or electricity generation. Therefore, for the whole lifetime of the vehicle and over the entire operating range, lower specific fuel consumption and reduced CO_2 emissions are achieved for the vehicles equipped with the diesel engine in comparison to its similarly rated spark ignition counterparts [5]. The issue of potential global warming demands reduction of green house gases. Diesel engine with its significant efficiency over gasoline engine can meet the low carbon emission requirement [6].

The difference between the engine built 100 years ago and the modern-day engine can be seen by comparing their thermal efficiency, reliability and emission level. Internal combustion engines are the dominant prime mover in several areas. Efficiency optimization and emission reduction are the main focus of development in internal combustion engines [7]. Increased use of technology such as engine downsizing, turbocharging, supercharging and variable valve timing is being extensively investigated to meet mileage requirement [1].

1.1 Turbocharged diesel engine

Continuous improvement in internal combustion engine technology is necessary to meet stringent performance and emission requirements. The developments in CI engine technologies have led to the increase in the demand of fossil fuels causing the rapid depletion of fossil fuel reserves [8]. Recent estimates for fossil fuel predict that the fossil fuels may be depleted in the coming 45 years [9]. Fuel saving in the transportation sector is a critical issue across the globe [10]. In addition, the transport sector is undergoing rapid transformation due to stringent emission regulations in most of the developed countries with a target of improving fuel efficiency of the vehicles [11, 12]. In order to reduce engine emissions, improvement in engine efficiency and combustion process technology is important to meet the increasing land transport requirements [13].

Trend of down-sizing is increasing in internal combustion engine development since last two decades [14]. Engine down-sizing is a key measure to reduce fuel consumption and lower exhaust emissions [15]. Engine manufacturers are forced to produce significantly down-sized engines due to the increase in fuel prices and the stringent emission regulations. However, the engine down-sizing results in lower power output. The output power can be improved by boosting or introducing extra air into the engine, resulting in burning more fuel to maintain specific air fuel ratio. Several methods of boosting are available such as superchargers, turbochargers or the combination of both can be used. Operating supercharged is a very favourable feature of diesel engine among its many merits. By supercharging, inlet pressure and density increases by means of a compressor located upstream of the engine. This enhances the air supply entering into the cylinder enabling efficient burning of higher amount of fuel and in turn increasing the engine power. A special case of supercharging is turbocharging which is the most preferred supercharging versions for many decades [16]. It is a promising way for energy saving and CO_2 reduction [17, 18, 19]. Turbochargers have the capability to enhance the engine power and torque without increasing the swept volume of each cylinder [20]. In a study [21], a comparison was made between turbocharged engine and naturally aspirated engine in terms of engine performance and emissions. While turbocharging decreases brake specific emissions, brake torque and brake power

1. INTRODUCTION

were boosted as compared to naturally aspirated engine as shown in Figure 1.2 and Figure 1.3 respectively.



Figure 1.2: Comparison of brake torque in turbocharged and naturally aspirated engine [21]



Figure 1.3: Comparison of brake power in turbocharged and naturally aspirated engine [21]

The reliability, widespread availability and excellent thermal efficiency of turbocharged diesel engine make them the prime mover for small and medium sized vehicles [22, 23].



A typical turbocharged diesel engine is shown in Figure 1.4.

Figure 1.4: Turbocharged diesel engine [24]

1.1.1 Geometry and performance of turbocharged diesel engines

The layout of turbocharging system is shown in Figure 1.5. Turbocharger is a device consisting of a turbine and a compressor attached to a single rotating shaft. In turbocharged diesel engines the heat that is generally lost in exhaust gases (around 30-40% in a naturally aspirated diesel engine [25, 26]) is recovered for compressing the fresh air into the engine cylinders. The exhaust gases are expanded in the turbine blades and part of the wasted energy is converted into useful work by the compressor. The compressor outlet is connected to the inlet manifold through an inter-cooler to increase the air density flowing into the intake manifold. In this way the volumetric efficiency of the diesel engine is enhanced by the use of turbocharger.

Turbocharger boosts the power output of the engine with the same displacement volume of the engine by providing intake air with increased density and hence increased mass. So, it is a boosting system enabling the engine to breathe more air resulting in more power output for a given size of the engine. A distinctive advantage of a turbocharged engine is that it operates more efficiently than its naturally aspirated counterpart, hence CO_2 production is proportionately less. This contributes to the increase in the production of turbocharged vehicles as concerns over greenhouse gas

1. INTRODUCTION



Figure 1.5: Layout of turbocharged diesel engine

emissions are now reflected in the automotive legislation also [16]. Turbocharging is considered as the most common boosting method used in the automotive industry.

1.2 Transient response of turbocharged diesel engine

Despite its several benefits, a serious operating disadvantage of turbocharger is its problematic transient behaviour which is due to the fact that at low loads and speeds the available exhaust energy is not sufficient enough to produce substantial boost [16].

There are essential operating discrepancies between the steady state and transient conditions of diesel engine. Crankshaft rotational speed, fueling and all other engine as well as turbocharger properties remain practically constant during steady state conditions. On the other hand, both engine speed and amount of injected fuel change continuously during the transient operation. The result is the variation of exhaust gas energy that affects the enthalpy of the turbine. Air supply and boost pressure are henceforth influenced through the turbocharger shaft torque balance [5]. Due to the often non-optimum performance of the diesel engine, the transient response is of crucial importance. Figure 1.6 shows the difference between steady state and transient operation of diesel engine in terms of engine's torque response. The torque response of naturally aspirated engine is linear. Contrary to this, the transient torque build-up of turbocharged diesel engine is highly non linear. The engine is unable to attain its



quasi steady state performance leading to torque deficit [16].

Figure 1.6: Comparison between steady and transient operation of diesel engine [16]

At high load transient conditions, faster response of the turbocharger is needed for the improvement in the drivability of diesel engines. However, turbochargers have their own inefficiencies and there are associated problems such as the turbo lag. The turbo lag refers to the time required for the exhaust system and turbocharger to generate the required boost. When a sudden load change is applied at lower engine speeds with rapid acceleration, the turbocharger must also accelerate to its new steady state. But the turbocharger doesn't respond effectively as compared to its naturally aspirated counterparts. Turbo-lag is the biggest disadvantage of turbocharged diesel engines due to which sufficient boost pressure can not be achieved at low engine speeds [27]. The reasons for turbo-lag are the inertia of the turbocharger rotor and the compressibility of exhaust gas within the engine as will be discussed in the next paragraphs.

A typical transient event of acceleration from low load and speed is described by Giakoumis [16] to highlight the concept of turbo-lag. With the fueling increase command, the fuel pump rack position changes instantly leading to higher fueling. At this moment higher amount of air is required to accommodate this elevated fueling. However, breathing is strongly influenced by the operation of the turbocharger in turbocharged diesel engine. Hence, the amount of air inducted into the cylinder is not only determined by the engine speed as in naturally aspirated counterpart but also by the

1. INTRODUCTION

compressor operating point. To illustrate the problematic behaviour of turbo systems, following equation can be referred which describes Newton's second law of motion for rotational system:

$$\eta_{\rm mtc} T_{\rm qt} - |T_{\rm qc}| = J_{\rm tc} \frac{dw_{\rm tc}}{dt} \tag{1.1}$$

In the above equation, $T_{\rm qt}$ and $T_{\rm qc}$ are the turbine torque and compressor torque respectively, $J_{\rm tc}$ is the turbocharger inertia, $\eta_{\rm mtc}$ is the shaft mechanical efficiency and $w_{\rm tc}$ is the turbocharger speed. From equation 1.1 it is clear that turbocharger's acceleration depends on turbine torque and inertia of the turbocharger. The turbine torque $T_{\rm qt}$ does not raise instantly due to two reasons: increased heat transfer lost to the cylinder and exhaust manifold walls and the subsequent filling of exhaust manifold with higher enthalpy gas. In addition, higher turbocharger inertia $J_{\rm tc}$ is the most influential factor that prohibits fast turbocharger shaft acceleration. Hence, during the early cycles the compressor operating point moves slowly towards increasing the boost pressure and air mass flow rate. The delay between fueling and air-supply is known as turbo lag and during this slow response period the engine runs with very limited boost. Consequently, the air fuel ratio becomes very low and the combustion deteriorates in diesel engines. This leads to slow engine response and increased exhaust emissions. One of the major challenges in turbocharged engines is achieving good transient performance and this is still being actively pursued. Combustion and emission performance of turbocharged diesel engines under transient operating conditions has been the focus of research in order to meet the increasingly stringent emission regulations [28, 29, 30].

1.3 Diesel engine exhaust emissions during transient operation

In addition to air-supply deficiency, the problematic transient response pattern gets amplified with in-cylinder combustion issues. As compared to steady state, boundary condition response delay under transient conditions is the characteristics of turbocharged diesel engine [31, 32]. The response delay causes poor in-cylinder physical condition leading to fuel economy deterioration and pollutant emissions [33, 34]. At the beginning of the transient event of speed or load increase, the higher pressure jets are injected into an environment that remains almost the same to the previous steady state conditions. The temperature adapts slowly to new loading conditions. Hence, liquid fuel impingement on the cool combustion walls increases that lowers the rate of mixture preparation. Subsequently, combustion gets deteriorated leading to increased exhaust emissions [16]. As compared to steady state operation, the combustion products are produced in an unpredictable manner under transient operating conditions [35].

Major emissions from internal combustion engine include nitrogen oxides (NO_x) , carbon monoxide (CO), hydrocarbons (HC), particulates (PM) and aldehydes. These products are the major source of air pollution. There may be many unregulated emissions as well. With the recognition of environmental issues such as impact of air quality on health, a great emphasis is made on the reduction of various emissions from the engine. Meeting the emission requirement is currently one of the major factors in designing and operation of the internal combustion engines [1].

1.4 Research problem and aims

Due to the fact that turbo-lag is the major reason for poor transient response of the diesel engine resulting in increased exhaust emissions [8], several technologies are developed over the years to cope with this problem. These technologies can be categorised as active and passive methods based on how the inappropriate air-fuel ratio is addressed. Quantification of turbo-lag for these technologies and its comparison for relative improvement in transient response to find the optimum technology is an important task which has hardly been investigated. Based on this, optimization of transient combustion and emission performance can be done which is essential to address the increasingly severe ecological issues in addition to the strict emission protocols [36]. With this motivation detailed investigations are performed in the present study to find the optimum technology and examine its effect on the overall performance of turbocharged diesel engine. The important aspects that needs to be addressed in the context of transient response of turbocharged diesel engine are described in the following.

In the event of rapid speed and or load changes, the operational capability of the diesel engine can be enhanced by improving the transient response of the diesel engine [37]. Since majority of daily driving schedules are transient in nature, hence the experimental and modeling analyses of transient diesel engine operation is an important aspect to be focused by the engine manufacturers. Numerical simulation are being used for several years to simulate the actual real working conditions in order to get more effective results. In actual physical tests, for example, intake manifold design and optimization, the trial and error method could be effective but very expensive and time consuming. Simulation technologies play a vital role in reducing time and cost as well as getting optimized design with little iterations [38]. Diesel engine modeling using

1. INTRODUCTION

advanced engine simulation software to simulate the effect of different technologies on the transient response of diesel engine can be a useful and reliable tool to compare them for optimum results which otherwise would be very difficult and costly using the real setups. This forms the first aspect of this research.

Rapid increase in fueling rate for turbocharged diesel engine represents a transient event [39] as due to this elevated fueling and insufficient air flow in the combustion chamber, air-fuel ratio becomes very low. Mostly transient events reported in literature are engine speed or load changes such as acceleration, load increase or even starting. Fueling rate change representing a transient event is never investigated before and so it can be a novel operating condition for which technologies can be tested for transient response improvement. Nevertheless, speed and load transients can't be neglected as these are the most popular transient events. Acceleration tests are the best representative of the transient operation of turbocharged diesel engine and turbo lag is more prominent during the acceleration from low engine speeds. Similarly, load increase transients are more difficult for the engine to cope with, as unlike acceleration, it initially causes drop in engine speed which poses another burden for the engine. Engine might even stall in circumstances where very high loading is applied instantly [16]. For these reasons it is worth evaluating the effect of optimum technology under these operating conditions and this forms the second research area.

In order to produce engines that meet stringent regulations concerning exhaust emissions, the study of transient emission is very important to the manufacturers [40]. For the overall performance of the engine, the effect of optimum technology should be evaluated for exhaust emissions to ensure that the chosen technology does not require compromise on emissions. This forms the third aspect of the current research.

A detailed literature review is performed in the next chapter on different methods of reducing the turbo-lag and improving the transient response. Specific research objectives will be discussed after identifying the knowledge gap through literature review. Based on the research problem, aims of the study are formulated below.

- 1. Development of engine model for the techniques of improving the transient response of turbocharged diesel engine and investigation on selecting the optimum technology based on turbo lag quantification.
- 2. To investigate the effect of selected technology on the transient response of diesel engine under various operating conditions.
- 3. To investigate the effect of selected technology on the environmental performance of turbocharged diesel engine.

1.5 Thesis structure

The entire thesis is organized into 7 chapters. Chapter 1 presents the overview of the turbocharged diesel engine and the basic information related to its transient response and subsequent exhaust emissions. Based on this overview, motivation for carrying out this research is described and aims of the study are formulated. Chapter 2 covers the literature review followed by specific research objectives. Chapter 3 examines the numerical/experimental methods employed to achieve the objectives. This section is broadly divided into two sub sections. Each sub section contains the details of the methodology employed. Simulation based investigations are carried out in Chapter 4 for the effect of passive methods of improving the transient response. Transient performance of the engine has been simulated through the rapid change in fueling rate. Parametric study has been undertaken to analyze the quality indicators. Turbo lag is quantified and taken as a measure of the transient response of the turbocharged engine. Chapter 5 provides experimental and simulation based studies on active method of improving the transient response and a comparison is made between active and passive methods. The potential improvement in the response of turbocharger is assessed through turbocharger response parameters and the technologies are evaluated for their relative contribution for transient response improvement. Based on maximum improvement in transient response, optimum technique is chosen. Detailed investigations are carried out to observe the effect of the selected technique on the transient response of the turbocharger under acceleration and load tests. In addition, the effect of selected technology on the environmental performance of the engine is reported in Chapter 6. Chapter 7 discusses the findings of the study and the contribution of the work is highlighted. Recommendation for the future work is also included.

1. INTRODUCTION

$\mathbf{2}$

BACKGROUND AND LITERATURE REVIEW

This chapter provides the background information on the transient response of turbocharged diesel engine, looking into the various methods of controlling turbo lag and hence improving the transient response of the turbocharger. Major exhaust emissions from the engine during transient conditions are also reviewed. The relevant literature is reported on the efforts made by the researchers to improve the engine response under different transient conditions affecting the overall performance of the engine. Finally the objectives of the study are presented based on the literature review.

2.1 Transient operating conditions

Turbocharged diesel engine still dominates the market of medium and large vehicles as well as ships with the advantages of high power and low fuel consumption. The speed of turbocharger can be determined by engine load and speed under steady state operation. However, in rapid changing conditions such as acceleration or load change, the turbocharger will show delay in response due to the inertia of turbocharger rotor and the volume of intake and exhaust pipes connected [41]. Great emphasis is made on transient operating conditions due to increasingly stringent fuel consumption and emission regulations [42].

Proper inter connection between engine, governor, fuel pump, turbocharger and load is important to address the non-optimum performance of turbocharged CI engine during transient operations. Diesel engine simulations have contributed significantly towards this aim [43, 44]. In terms of engine response and in the reliability of governers and fuel pumps, rapid load changes are very demanding.

Transient load of a diesel engine can be described as either speed variation under constant torque or variation in torque under a constant speed or simultaneous changes in both. The effect on system response, emissions and fuel efficiency are of main interest to researchers and developers [45]. The transient operating conditions include [22]:

- 1. Changing torque and changing speed decreasing speed and decreasing torque (DSDT), decreasing speed and increasing torque (DSIT), increasing speed and decreasing torque (ISDT), increasing speed and increasing torque (ISIT)
- 2. Constant speed and changing torque constant speed and increasing torque (CSIT), constant speed and decreasing torque (CSDT)
- 3. Constant torque and changing speed constant torque and increasing speed (CTIS), constant torque and decreasing speed (CTDS)
- 4. Starting
- 5. Parking

As per the studies conducted by Qiang Liu et al.[22], transient operation under constant speed and changing torque is most popular due to its higher weight (61.13%) as compared to other conditions, weights assigned based on the transient characteristics and some assumptions by He [46]. Moreover, smoke emission degradation is observed at CSIT rather than CSDT, so CSIT is the main focus for the investigations under load transient in the present study.

Rakopoulos and Giakoumis [47] presented detailed analysis of the operation of an indirect naturally aspirated diesel engine under various load conditions. These include - load intensity, load schedules and load types. The results show that the speed droop and recovery time were seriously affected by the load intensity as well as by the load types. However, they were not significantly effected by the load schedule. The term speed droop meant maximum deviation of speed from its initial value divided by the initial speed. Recovery period was the time needed for the engine to reach 98% of its final speed. These results are shown in Figure 2.1 and Figure 2.2. Figure 2.1 shows that greater load leads to lower final speed. However the recovery time was not much affected for final load greater than 70%. The effect of load schedule is shown in Figure 2.2. Although the final speed and recovery period were not significantly affected yet faster load application shows lower final speed and recovery period.



Figure 2.1: Effect of load change on final speed and recovery period [47]



Figure 2.2: Effect of load schedule on final speed and recovery period [47]

2.2 Existing methods for controlling turbo lag

In turbocharged diesel engines the problem of turbo lag arises during the transient operating conditions. This problem arises from the lack of air induction due to the slower response of the turbocharger during a transient. To meet the requirement of complete combustion, sufficient air should be supplied to the engine. Hence a good air injection system is important to achieve higher diesel engine performance [48]. For sudden load application in turbocharged diesel engine, the transient performance of the engine can be improved by controlling the fuel flow and by accelerating the turbocharger by some external means [49]. Control of fuel flow can be achieved through the use of boost pressure sensitive fuel limiters that limits the fuel flow until the boost pressure achieves a value so that very low air fuel ratio is prohibited. Soot emissions are reduced by this method, however, the drawback is slower engine response and poorer vehicle drivability. So, controlling only the fuel flow can not be the sole measure during transients rather turbocharger specific measure should be taken to address the transient delay due to slower turbocharger response. Six such measures are reported by Giakoumis [16], which are listed below:

- 1. Lower-inertia turbocharger
- 2. Variable-geometry turbine (VGT)
- 3. Combined supercharging
- 4. Two-stage turbocharging
- 5. Sequential turbocharging
- 6. Electrically turbocharger assist (ETA)

The above listed methods are the indirect method of improving the transient response achieved through faster turbocharger response to provide sufficient boost in order to meet the elevated fueling during transients. Such methods are referred as passive methods. Another method is active method of meeting the air-supply demand by means of air-injection as reported in [49, 50, 51].

Figure 2.3 shows *Combined supercharging* in which two compressors are arranged in series realized through the use of a positive displacement compressor which is located upstream of the turbocharger compressor. At low engine load and speeds when the enthalpy capacity of exhaust gases leaving the cylinder is low and the turbocharger is incapable of producing the required boost, the positive displacement compressor

becomes active at this moment and provides sufficient boost thereby improving the transient response. Nevertheless, the drawback of this configuration is reduced engine efficiency owing to the extra power needed to drive the compressor [16].



Figure 2.3: Combined supercharging [16]

In *Two-stage turbocharging* as shown in Figure 2.4, two units (turbochargers) as low pressure (LP) and high pressure (HP) are connected such as LP unit is fitted upstream of the HP one with an inter cooler between the two compressors. High compression ratios are employed by this method and efficiency is increased as greater fuel saving is achieved due to elevated boost pressure. But two issues need to be addressed for this configuration - Firstly, due to highly changing operating conditions, a control system is required. Secondly, engine and turbocharger matching is more difficult due to the added complexity by the second turbocharger [16].


Figure 2.4: Two-stage turbocharging [16]



Figure 2.5: Variable geometry turbine (VGT) [16]

VGT is shown in Figure 2.5. The turbine area gets adapted according to the continuously changing engine's operating conditions. During the transient period when the boost pressure is typically low, the turbocharger control system reduces the turbine area by closing down the vanes. This in-turn increases the pressure and temperature of the turbine resulting in increased turbine torque. Thus according to equation 1.1, higher acceleration of turbocharger shaft is achieved that promotes faster transient response. However, extensive matching procedure is needed in addition to sophisticated control system to determine the exact opening and closing pattern according to engine and turbocharger parameters.

In a recent study by Giakoumis and Tziolas [52], a GT-Power model is constructed for a variable geometry turbocharged (VGT) truck diesel engine. Turbocharger as well as engine performance parameters are investigated. The study highlights the benefit of VGT as compared to fixed-geometry operation in terms of higher boost pressure and air supply to the engine cylinders. An instrumental test bed was set up by Rakopoulos et al. [53] to investigate the key engine and turbocharger parameters. A variety of starting conditions are taken into consideration. The results indicate that turbo lag is the major contributor for increased pollutant emissions in all the cases.

Two-stage turbocharger and VGT are compared by Choi et al. [54] and they repoerted faster response at 1000 and 2000 rpm by using two-stage turbocharger as compared to VGT. Further, two-stage turbocharger achieved higher torque at low engine speed as compared to VGT.

In Sequential turbocharging configuration (shown in Figure 2.6), two or more turbocharger are arranged in parallel which is in contrast to two-stage turbocharging. In this configuration also smaller units for faster response are connected as in twostage turbocharging. Sequential turbocharging configuration is complex and accurate matching process is required to avoid the risk of surge when a unit is engaged and disengaged [16]. As per the studies conducted by [55] and [56], the things become more complicated if aftercooler(s) and VGT are incorporated in sequential turbocharging or when more than two units are connected in sequential turbocharging.

ETA is shown in Figure 2.7, an electric motor/generator is used to assist the turbocharger operation. ETA device are also referred as hybrid turbocharger or turbocharger motor/generator [57]. For the reduction of turbo lag during transient operations, electrical hybridization of turbocharger has gained widespread attention as it has potential to be an effective method of improving air delivery into the engine. The motor/generator is added to the turbocharger which can be integrated in various different arrangements. The scope of electrical turbocharger assistance is reduction in time to boost the pressure by the application of the motor/generator which increases the power available to accelerate the rotor assembly [58].



Figure 2.6: Sequential turbocharging [16]



Figure 2.7: Electric torque assist (ETA) [16]

Among the passive methods inertia reduction using lighter materials and ETA are still promising options to improve the transient response. Furthermore, it is worth considering active method of air injection also for review in view of improving the transient response. Hence in the next sub section, review will be carried out on the three main technologies for the reduction of turbo lag: Electric torque assistance, air injection system and by the method of reducing the inertia of the compressor

2.2.1 Electric torque assistance

Tavcar et al. [59] compared the performance of a baseline turbocharged high speed direct injection engine with different types of electrically assisted turbochargers (EATs) as given below:

- Electrically assisted turbocharger (EAT)
- Additional electrically driven compressor (TEDC)
- Electrically split turbocharger (EST)

Analyses were carried out for different driving conditions, including tip in the gear and the new European drive cycle. They found that all EAT topologies improved the transient response of the engine and the driveability of the vehicle. These topologies could also improve the steady state torque output of the engine with retained fueling.

Results reveal that an engine utilizing TEDC delivers the highest torque output and fastest transient response at low engine speeds. While at high engine speeds, EAT provides high torque output in addition to the transient response improvement. EST provides the same transient response as that of EAT, since both are constrained by compressor surge limit. For urban driven cycles, the utilization of EST enables the reduction in fuel consumption. The study concluded that the ideal arrangement of electric turbocharger depends mainly upon the application [60]. The major disadvantage of such systems is increased inertia and their inability to meet low packaging space requirements in modern engines.

The simulation results obtained by Ibaraki et al. [61] reveal 50% improvement in engine torque by a 1 kW motor assist while 100% improvement was achieved with a 2 kW motor assist. Furthermore, the fuel efficiency improved by 8% with a 1 kW motor assist while 12% improvement in fuel efficiency is observed by 2 kW motor assist [61]. Similar investigation performed by Cieslar [50] on time to achieve torque rise shows better performance by the 2 kW motor in comparison to 1 kW motor. However, the 3 kW motor gives very little additional benefit. These studies suggest that the assistance provided by the 2 kW motor is the optimum and indicate that it is equivalent to 0.16 Nm of additional torque [50, 61]

2.2.2 Inertia reduction of the turbocharger

Decreasing the turbocharger's mass moment of inertia plays important role for faster turbocharger and in-turn engine response as clear from equation 1.1. There are following three basic approaches reported in [16] for reducing turbocharger's mass moment of inertia:

- 1. The most straightforward is through employing lighter materials in order to decrease the turbine rotor mass. For this purpose silicon nitride and ceramic materials are the two successful choices.
- 2. Rather than using a single turbocharger, two units can be installed with smaller diameters. By the dependency of inertia on the fifth power of rotor diameter [62], this reduces the inertia of the system.
- 3. Smaller frame turbocharger can be utilized for lower inertia values.

In case of smaller frame turbocharger, efficiency decreases at higher engine speeds. Overboosting at higher engine speed and load is another adverse effect associated with smaller turbocharger frame which could be dangerous for safe engine and turbocharger operation [16].

In a study by Gilkes [63], different inertia values were taken as a percentage from the original compressor. Ricardo Wave simulation software was utilised in the study to investigate the effect of reducing inertia. The pressure response of turbocharger with different inertia values was plotted and the result shows that delay in the turbocharger response was reduced by reducing the inertia of compressor. Moreover, they compared the active method of injecting air into the inlet manifold and passive method of reducing the inertia of the compressor wheel and the results show that both methods reduce turbo lag, however the active system produces superior results. Turbo lag reduction achieved by inertia reduction method was found to be very small.

2.2.3 Air injection strategies

During the operation of CI engines, the air fuel ratio should be maintained within tight limits to obtain maximum engine output. During the turbo lag this ratio reduces and hence the quantity of air available may not be sufficient to meet the torque demand. During the period of low speed and fast acceleration, the air can be injected using different strategies to improve the air supply during transient operation of the turbocharger. Lee and Choi [51] performed detailed experimental study to improve the low speed torque and acceleration performance of the turbocharged engine system by injecting air into intake manifold during transient conditions. They found that the air injection into the intake manifold during the rapid acceleration greatly improves the combustion performance of turbocharged diesel engine.

Air injection is an effective technique to improve the transient response of marine diesel engine [48]. When a ship encounters a sudden danger in the sea, effective stopping of the ship is required followed by acceleration in the reverse direction rapidly. Turbocharged diesel engine is considered as a prime mover of a ship and hence the success of the ship's emergency stop relies on the improvement of the rapid acceleration after the engine is put in reverse. The response of the turbocharger under the transient event of rapid acceleration plays an essential role in this situation. Wei [48] employed air injection technique to improve the turbocharger response. Air injection was applied at the compressor exit. Injection timings and durations were varied and compared under different working conditions. The results of the study demonstrate that additional air injection helps to stop a marine engine faster in case of sudden danger.

Gilkes et al. [64] analyzed the effect of injecting air at either an inlet manifold and compressor or both the manifolds and compressor which is also known as two-point injection system. They performed a parametric study analyzing four parameters that are vehicle speed, outlet pressure of compressor, turbine inlet temperature and driver demand. It is found that the transient response of the turbocharger as well as the overall performance of the engine was improved using air injection systems. With air injection system the turbo lag is reduced by 71% in achieving the vehicle response of 30 km/hour. However, two-point air injection has shown small improvement.

Ceislar [50] examined the effect of air injection at various locations within the engine system and compared the results to establish the optimal injection location. The results revealed that air injection at exhaust manifold overcomes the problem of compressor surge while improving the transient response. In this investigation, the air injected at exhaust manifold is adjusted by a control valve and is stored in a compressed gas tank. The gas tank is recharged during the engine overrun when the pressure in the exhaust manifold is high. So, the compressed gas is collected in the tank during braking to be used later for assistance during a tip-in (acceleration phase). The system design for this functionality is very simple as it requires only a compressed gas tank, an additional exhaust manifold connection and a control valve. Standard diesel EGR/VGT is included in engine air path for the Braking Exhaust Energy Storage (BREES). The transient response of BREES is analyzed when the turbo lag is more pronounced i.e. in high gears. It is worth noting that the time to torque during 3rd gear tip-in acceleration is reduced by 60%. However, the recharging of Braking Exhaust Energy Storage (BREES) is slow, demanding the clutch to be disengaged for about 3 seconds. Also the volume of gas tank is relatively larger for passenger cars.

The pros and cons of various methods of improving the transient response as discussed above are summarized in Table 2.1.

Technique	Pros	Cons	
Combined	Improved transient response by		
supercharg-	means of positive displacement com-	Reduced engine efficiency	
ing [16]	pressor		
Two stage	Increased officiency with less fuel	Matching difficulty between the two turbochargers, control system re- quirement	
turbocharg-	increased enciency with less fuel		
ing [16]	consumption		
Variable	Increased turbing torque factor	Requirement of highly conhistigated	
geometry	transient response	control system	
turbine $[52]$	transient response		
Sequential		Complex configuration, accurate	
turbocharg-	Faster response	matching process was required to avoid surging	
ing [16]			
Electric	Reduction in time to boost the pres-	Ingrossed inertia and more space re	
torque as-	sure, power available to accelerate	increased inertia and more space re-	
sist $[57]$	the rotor assembly was increased	quired to instan the motor	
Inertia reduc-			
tion of the	The delay in turbocharger response	Limited literature is available on in-	
compressor	was reduced	ertia reduction	
wheel [63]			
Ain injustion	Improves the transient response of	The effect of air injection on arris	
Air injection	turbocharger as well as the over all	ine enect of air injection on emis-	
system [51]	performance of the engine.	sions was not considered.	

Table 2.1: Various methods of improving the transient response with their pros and cons

A common rail direct injection (CRDI) engine was taken into consideration for air flow rationalization as well as turbo lag reduction [65]. The authors proposed a strategy for air injection directly into the combustion chamber. Accumulator and pressure regulator valve are the two additional components in the proposed system. At high engine speed and low torque demand conditions, the pressure regulator allows the high pressure air in the inlet manifold to flow into the accumulator, to be stored there for later use during the turbo lag. So, during low inlet manifold pressure when the torque demand is high, the pressure regulator valve allows the air flow from the accumulator to the inlet manifold. Hence the accumulator supplies the air according to the torque demand and balances acceleration and deceleration.

2.3 Diesel engine exhaust emissions under transient events

To meet the stringent emission regulations, it is important to improve the emission performance of the diesel engine. Like most of the fossil fuels, diesel fuel is mainly composed of carbon and hydrogen. Only CO_2 and H_2O would be generated by the combustion process in an ideal thermodynamic equilibrium. However, many harmful products such as CO, HC, NO_x and PM are generated during combustion due to the following reasons [66]:

- Air fuel ratio
- Ignition timing
- Turbulence in the combustion chamber
- Combustion form
- Combustion temperature etc.

The incomplete combustion leads to CO emission. The concentration of CO largely depends on the air/fuel mixture. For a rich mixture the excess air factor (λ - which is the ratio of stoichiometric air fuel ratio to actual air fuel ratio) is less than 1. Due to air deficiency and reactant concentration in the rich mixture, all the carbon cannot convert to CO₂ and CO is formed. Formation of CO in diesel engine is minimum because they are lean combustion engines having high air fuel ratio ($\lambda > 1$). However if more droplets are there in the engine and if insufficient turbulence or swift is generated CO is produced. CO₂ emission is considered as the main contributor of global warming. The combustion of carbon in the fuel results in CO₂ emission [67]. In the presence of adequate oxygen and high temperature of the gases, CO may get oxidized to CO₂.

Highly compressed hot air is used in diesel engine to ignite the fuel. Due to very high temperature (above 1600 °C) in the cylinders, nitrogen reacts with oxygen and generates NO_x (nitrogen oxides) emission. Therefore temperature and concentration

of oxygen in the combustion are the main factors for the formation of NO_x . This is illustrated in Figure 2.8 [40]. 10-80% load increase event was taken into consideration. The Figure shows the development of NO concentration in the exhaust along with the maximum gas temperature and oxygen concentration in the fuel spray. In the beginning when the load is low, NO emissions were very low. With the increase in load, NO emissions were also increased due to its dependence on temperature. Diesel engines need higher temperature than gasoline engines because they are compression ignition engines. Therefore diesel engines are the main contributors to NO_x emissions. Nitrogen oxide (NO) and nitrogen dioxide (NO₂) are together called as nitrogen oxides. NO gradually converts to NO₂ at atmospheric pressures.



Figure 2.8: NO concentration in the exhaust, maximum gas temperature and oxygen concentration in the fuel spray [40]

Fuel evaporation from the open areas and the incomplete combustion leads to the formation of hydrocarbons [67]. Although diesel engine for automobile has the advantage of superior dynamic and economy [68] but NO_x emissions are high which is a threat on human health and environment [69, 70]. Where as CO, HC and greenhouse emissions are low. Relatively high NO_x emissions in diesel engine is because of high combustion temperatures and lean air-fuel mixtures [71]. The engine exhaust typically

contains 70-90% NO and 10 - 30% NO₂ [72].

Emission directives are published by European Commission for engine manufacturers regarding emission requirements. 'Euro 6' standard came into picture in January 2015. Since more emissions are produced by diesel engines, therefore it becomes difficult for engine manufacturers to adhere to these emission regulations. Maximum limit by Euro 6 for NO_x is 80 mg/km and 170 mg/km for hydrocarbons and nitrous oxide emissions. In addition to being very expensive, it is technically challenging to produce a vehicle that adheres to the Euro standards [73].

In a study [74], the effect of air injection by an air cell on the combustion and emission characteristics of a direct injection diesel engine was experimentally demonstrated. The study concluded that the use of an air cell in engine improves the pollutant emissions while BSFC remains constant. In another study [75], researchers experimentally studied the effect of establishing an air-cell inside of the piston body at different injection timings and observed the soot, NO_x emissions, combustion and performance parameters. They reported simultaneous reduction in soot and NO_x emissions in addition to engine performance improvement. Zhou et al. [76] studied heavy duty truck engine with an intake valve closing mechanism and a two-stage turbocharger. They highlighted transient emissions spikes reduction. In an experimental study [77], the effect of intake throttle valve opening on NO_x emissions under world Harmonized transit cycle (WHTC) was examined and 41.5% reduction was observed in NO_x emissions while the fuel economy increased slightly. Guan et al. conducted an experiment on a HD diesel engine using exhaust gas recirculation (EGR) and Miller cycle and they reported minimum impact on fuel economy and smoke emission while lower engine-out NO_x emissions and higher exhaust gas temperatures (EGT) were achieved [78]. Strict regulations on emissions demand the implantation of after treatment devices along the exhaust line of diesel engines [79]. By means of boost and EGR control the lack of temperature at the inlet of the turbine can be overcomed leading to power management in the exhaust gas [80].

Fuel efficiency and NO_x emissions were analyzed under cyclic transient loads for a turbocharged diesel engine by Yum et al. [45]. Load profile consisting of a series of sinusoidal wave with a fixed amplitude at different frequencies and mean values were applied. They reported the average effect of cyclic load on the fuel consumption and NO_x emissions. The effect depends upon the average load level and as the load level was lowered, it became more visible.

2.4 Summary and knowledge gap on methods of controlling turbo lag

In most of the passive methods reviewed above for improving the transient response, more than one unit is combined to achieve lower turbocharger inertia. These include Combined supercharging, Two-stage turbocharging and Sequential turbocharging. As discussed in the previous paragraphs, these configurations require compromise in terms of efficiency, matching and sometimes fuel consumption as in case of combined supercharging. In case of VGT, the main drawback is reduction in the efficiency of engine and/or turbocharger as the turbine can not operate efficiently throughout the whole operating range [16].

Electrically assisted turbochargers are widely used for the improvement of transient response of the engine and hence drivability. Findings of simulation based investigations using software described in the literature are very important to model the optimal real systems before its implementation. As discussed before, rapid increase in fueling rate is rarely investigated before. Electric torque assistance will be simulated using Lotus Engine Simulation under the event of fueling rate change and its effect on transient response characteristics will be reported.

Furthermore, although it is well documented that inertia reduction of the compressor is one of the promising method of improving transient response of turbocharged diesel engine, nevertheless very limited literature is available on the investigations performed using inertia reduction. In the study [63], effect of inertia reduction is observed only on compressor outlet pressure. Compressor speed is another important parameter which represents the transient response of the turbocharger, as during the transient the compressor doesn't accelerate appropriately to meet the air supply demand. Moreover, since reductions in inertia are difficult to achieve, hence it is important to perform further studies which could be using different simulation program. In addition to benchmark the results of [63], the effect of inertia reduction on compressor speed will provide useful insight on the transient response improvement. Use of Lotus engine simulation for this purpose will be a new study to model this effect.

Investigations on air injection technique reported in the aforementioned review are under taken for light duty and medium duty vehicles. However, the effect of air injection on the response of heavy duty vehicles is scarcely reported. There is essential difference between the passenger cars and heavy duty vehicles. The effect of air injection on heavy duty vehicle is considered for marine applications only [48]. But the working conditions of marine applications and road vehicles are quite different. Marine engines normally run at steady state conditions for most of its service. The load remains constant and speed is also relatively fixed while the speed and load of automobile engines keep changing depending upon the conditions on the road. This indicates the need for considering the effect of air injection on heavy duty turbocharged road vehicles [50]. Furthermore, as reported in the review on air injection, for the realization of air injection pressure a compressed gas tank is required which is difficult to incorporate in passenger cars. While in heavy duty vehicles it is not challenging. This is another reason for choosing heavy duty engine used in road vehicles for investigations on air injection.

Use of fossil fuels in automotive sector is one of the primary causes of greenhouse emissions. The automotive engines need to perform at their best efficiency point to limit these emissions. Most of the quality indicators in this regard are based on near steady state global operational characteristics for engines without considering local performance [81]. Considerable amount of research has been carried out as reported in [40] on modelling the thermodynamic and gas dynamic processes as well as combustion in diesel engines. However, exhaust emissions under the transient operation is investigated mainly on experimental rather than simulation basis due to high computational times required for the analysis of each transient cycle. Only few studies [73, 82, 83] are conducted predicting the exhaust emissions through modeling. Furthermore, the techniques employed for transient response improvement are rarely investigated to reduce the exhaust emissions during transient operation. It is important to evaluate the technique used for improving the transient response for exhaust emissions.

2.5 Aims and objectives of the study

Based on the three research problems identified through literature review, the overall aim of the project is to investigate the techniques for improving the transient response of turbocharged diesel engine through modelling and making sure that the technology doesn't need compromise on emissions. The research aim can be achieved by addressing the following objectives:

- **Objective 1:** Engine model development to quantify turbo-lag through torque assistance
- **Objective 2:** Engine model development to quantify turbo-lag by reducing the inertia of rotating component.
- **Objective 3:** Engine model development to quantify turbo-lag through air injection at intake manifold.

2. BACKGROUND AND LITERATURE REVIEW

- **Objective 4:** Optimization of transient response improvement technique by comparing the three different methods of turbo lag reduction under the event of fueling rate change.
- **Objective 5:** Evaluation of the effect of selected technology for acceleration tests.
- **Objective 6:**Evaluation of the effect of selected technology for load magnitude variation.
- **Objective 7:** Evaluation of the effect of selected technology for load schedule variation.
- **Objective 8:** Evaluation of the effect of selected technology on emission characteristics of the engine under speed transient.
- **Objective 9:** Evaluation of the effect of selected technology on emission characteristics of the engine under load transient.

3

ENGINE SIMULATION

Calculation models are the research tools not only to guide the expensive experimental studies towards optimal solution for engines but they can also generate more information than experiment on the physical phenomenon taking place in the engines [84]. Computer simulations play important role in predicting the performance of engine systems. For studying the engine performance, diesel engine simulation modeling is an effective tool that contributes to the evaluation and new development in this field [85]. Furthermore, since turbocharger is a vital part of most modern diesel engines, the importance of turbocharger simulation can't be neglected. The simulation of turbocharger can provide accurate predictions of performance and emissions of turbocharged diesel engines [86].

Currently, within the engineering sector in particular, computer simulation is gaining widespread attention to model complex real world problems which otherwise require expensive experimental setups. Several commercial engine simulation packages are available in the engineering industry related to the design and optimization of internal combustion engines. The four engine simulation commercial packages most commonly used in the automotive industry are Ricardo Wave (RW), Lotus engine simulation software (LESoft), AVL Fire and GT-Power. These simulation software are similar in purpose and functionality. In a study performed by Chan et al. [87] a comparison was made for Ricardo Wave and Lotus engine simulation. These two software are tested on engine control capabilities. The result of the study reveals that the two codes predict similar pressure and temperature profiles in the engine. Figure 3.1 shows the flow diagram of the simulation model used in the modeling studies. The simulation modeling consists of preprocessing, run and postprocessing. In preprocessing, simulation model is constructed using different components such as engine, turbocharger,

3. ENGINE SIMULATION

intake valves, exhaust valves and fuel injectors. Strategies are developed for different tests and a control system is formulated to model a specific problem. The model is then run and the results obtained in postprcessing are analyzed.



Figure 3.1: Steps in simulation modelling

The present study is performed in two stages. Initially the author could only avail the Lotus Engine Simulation software. So, in the first stage based on the literature review on the capabilities of the software, LES is used to simulate electric torque assistance and further to investigate the effect of reduction in inertia of compressor wheel. These investigations cover the objectives 1 and 2 respectively.

In the next stage Ricardo wave is used to investigate the effect of air injection systems. Switching to Ricardo is done because of its flexibility for sub models, efficient calculation and well tested transient simulation capabilities. The Ricardo wave model is validated against the data collected from the real CI engine and after validation detailed investigations on air injection system are carried out to satisfy objective 3 to objective 9.

In LES software the engine simulation is processed in two modules named as the data module and the solver module. The user inputs the engine data through the data module. The solver module involves combustion and heat transfer zero-dimensional equations and fuel/gas composition solver based on the data provided in data module. The code can predict the gas flow, combustion and overall performance of internal combustion engines. Ricardo Wave analyzes the dynamics of pressure waves, mass flows, energy losses in ducts, plenums and manifolds. It provides time dependent fluid dynamics simulation and thermodynamics using two-zone model [59].

3.1 Lotus engine simulation

The engine model is developed using Lotus Engine Simulation software which can predict the complete performance of an engine system. The software provides a Concept Building Tool to build a complete engine model based on these three key parameters: number of cylinders, swept volume and engine speed at maximum power. As presented by Pearson et al. [88], the engine power is given by

$$W = \frac{P}{1200} V_{\text{swept}} N \tag{3.1}$$

where P is the brake mean effective pressure (BMEP) in bar, V_{swept} is the engine swept volume in litres and N is the engine speed in rev/min. The primary factor for determining the power output from an engine of a given swept volume is its maximum rated speed. Hence the maximum rated speed together with the number of cylinders and swept volume is responsible for defining the engine performance. The BMEP at a fixed air-fuel ratio (AFR) and given fuel specific heat value (Q_v) is related to the volumetric efficiency (η_v), thermal efficiency (η_{therm}) and the combustion efficiency (η_{comb}) by the equation:

$$P\alpha \frac{Q_v}{AFR} \eta_v \eta_{therm} \eta_{comb} \tag{3.2}$$

Lotus Engine Simulation uses a 'heat release' model to simulate the energy release mechanism from the mixture. In this model the mass fraction of burn fuel is calculated using a Wiebe function [89]. To model the combustion inside the cylinder, two-part Wiebe function is utilized. It has two consecutive combustion periods: premixed and diffusion. For predicting the burned mass fraction in each combustion period, following equations are used [90]:

$$m_{\text{premix}} = 1 - \left[1 - \left[\theta/\theta_{\text{b}}\right]^{\text{C}_{1}}\right]^{\text{C}_{2}}$$
(3.3)

$$m_{\rm diff} = 1 - e^{-\mathcal{A}(\theta/\theta_{\rm b})^{\rm M+1}} \tag{3.4}$$

In the above equation, C_1 , C_2 , A and M are Wiebe coefficients. θ is the burn angle and θ_b is the total burn angle. Heat transfer to the three major surfaces cylinder head, piston and liner is evaluated from the difference between the instantaneous gas temperatures and the metal surface temperature. Annand heat transfer model is used to calculate cylinder heat losses [39]. The convective heat transfer coefficients are derived from the well- known semi-empirical relationships of Annand model. Simulation of the energy release rate and the heat transfer processes enables to predict the cylinder pressure during combustion and hence a complete cycle simulation is performed [88]. Annand model is defined as:

$$ARe^{\rm B} = hB_{\rm cyl}/k \tag{3.5}$$

where A and B are the Annand coefficient for open and closed cycles, B_{cyl} is the cylinder bore, h is heat transfer coefficient and k is the thermal conductivity of the gas in the cylinder. Further, following equation determines the heat transfer per unit area [91]:

$$dQ/A = h(T_{\rm g} - T_{\rm w}) + C(T_{\rm g}^{4} - T_{\rm w}^{4})$$
(3.6)

In the above equation, T_g is the temperature of gas, T_w is the temperature of cylinder wall and C is the Annand coefficient only for closed cycle.

As the turbocharger is used to deliver more power out of a smaller engine, a relatively smaller engine is chosen to be modeled. A four cylinder, four stroke, CI engine is modeled in this study. The engine specifications are shown in Table 3.1. It shows the specification of turbocharged diesel engine based on engine type, stroke, bore, number of cylinders, combustion system, compression ratio, connecting rod and displacement. The specifications match with the specifications of the engine used in a mid range car as mentioned in [81]. A fully connected turbocharged model is shown in Figure 3.2. Engine cylinders, components of intake and exhaust manifolds, sensors, actuators and turbocharger are explicitly demonstrated on the figure. The manifolds have been modeled using plenum, pipes, valves and ports (blue at intake and orange at exhaust). Cylinders specifications are provided as shown in Table 3.1. Bore is taken as 88.0 mm, stroke as 85.0 mm, connecting rod length as 144 mm and compression ratio as 19.5:1. Firing order is set as 1-3-4-2 where cylinder 1 is the top cylinder. Valves are selected from the intake component tab and attached to the cylinder elements via connectors. Inlet and exhaust boundaries are added to complete the basic model. Fuel element is added to the model and fuel type is selected as diesel and the fuel system is chosen as direct injection. Compressor is connected to the inlet plenum and the turbine is connected with the exhaust manifold.



Figure 3.2: Turbocharged engine model in Lotus Engine

Engine Type	4 Stroke Diesel
Stroke	$85 \mathrm{mm}$
Bore	88 mm
Number of Cylinders	4
Combustion System	Direct Injection
Compression Ratio	19.5:1
Connecting rod	144 mm
Firing order	1-3-4-2

 Table 3.1: Engine specification

3.2 Ricardo wave simulation

Currently, within the engineering sector in particular, computer simulation is gaining widespread attention to model complex real world problems which otherwise require expensive experimental setups. Ricardo Wave is a 1 D engine and gas dynamic simulation software package commonly used worldwide. In this model, the combustion process is modeled by Diesel Weibe combustion model, which is an appropriate heat release model. The model uses functions to determine the correlations for premixed and diffusion burn with addition of tail burning which represent the slow late burning. The mass fraction burned, W is given by the equation [92]:

$$W = P_f [1 - [1 - (0.75\tau)^2]^{5000}] + d_f [1 - [1 - (cd_3\tau)^{1.75}]^{5000}] + t_f [1 - [1 - (ct_3\tau)^{2.5}]^{5000}]$$
(3.7)

where

 P_f - mass fraction of the premix

 d_f - mass fraction of the diffusion

 t_f - mass fraction of the tail burn curves

 cd_3 - coefficient of burn duration for the diffusion

 ct_3 - coefficient of burn duration for the tail burn curves

The following equations are used to find d_f , t_f , cd_3 and ct_3 :

$$d_f = 1 - P_f(1 - \alpha) \tag{3.8}$$

$$t_f = 1 - P_f(\alpha) \tag{3.9}$$

where

$$\alpha = 0.6 * [min(\phi, 0.85]^2 \tag{3.10}$$

$$cd_3 = \frac{0.055}{1 + 0.5min(\phi, 0.85)} \tag{3.11}$$

$$ct_3 = \frac{3.7cd_3}{1.12min(\phi, 0.85) + 1} \tag{3.12}$$

In the above equations, ϕ is the equivalence ratio.

Finally, the equation used to determine the burn duration term:

$$\tau = \frac{\theta - \theta_b}{125(\frac{RPM}{BRPM})^{0.3}} \tag{3.13}$$

where

 $\theta = \text{Crank angle}$

 $\theta_b = \text{crank}$ angle at the start of combustion

RPM = Engine speed

BRPM = User-entered reference speed

For heat transfer process, the Woschni correlation model is used. The Woschni model expresses the convective heat transfer coefficient as given by the equation [92]:

$$h_q = 0.0128 D^{-0.20} P^{*0.80} T^{0.53} V_c^{0.80} C_{enht}$$
(3.14)

where

D: Cylinder bore

P: Cylinder pressure

T: Cylinder temperature

 V_c : Characteristic velocity

 C_{enht} : User entered multiplier (For diesel jet combustion sub-model $C_{enht} = 1$)

In the present study a four cylinder, four stroke, turbocharged diesel engine available at the Advanced Automotive Laboratory of University of Huddersfield, UK is used to validate the engine model constructed using Ricardo Wave simulation software. The description of the diesel engine is provided in Table 3.2. The engine specifications provided are applicable to the heavy duty diesel engine [67].

Different parameters in Ricardo wave package are modelled using a number of mathematical as well as empirical formulae. For heat release, Diesel Wiebe function [89] is used as due to the unavailability of actual heat release profiles, this is the most appropriate heat release model. Model parameters are used to adjust the burn duration over the engine speed range. An ignition delay model is included in the Wiebe function to define the crank angle at which combustion is initiated. Woschni model for convective heat transfer is used to calculate the heat transfer [89]. In this model, simple heat transfer is assumed from a confined volume surrounded by all sides by walls representing the cylinder head, cylinder liner and piston face areas exposed to the combustion chamber [64].



Figure 3.3: Turbocharged engine model in Ricardo Wave

Wave Build is a GUI interface to build a wave model. A fully connected turbocharged engine model is shown in Figure 3.3. Elements representing different engine components are connected by ducts. Cylinders are selected from the flow element and as per Table 3.2 bore is taken as 103 mm, stroke as 132 mm and compression ratio as 18.3:1. For intake and exhaust valve, the basic valve parameters are entered as shown in Table 3.3 [93]. The intake and exhaust manifolds are modeled using ambients, ducts, orifices and junctions.

From the engine general panel, engine type is chosen as diesel and the specifications are entered as described in Table 3.2. As discussed earlier, Diesel Weibe is chosen as the combustion model and Woschni model is chosen for heat transfer. Fuel injectors are added to the model. From the turbo junction flow element, compressor and turbine are added on the canvas and a turbo shaft is selected from the mechanical element to connect the compressor and turbine. Compressor is connected to the intake manifold and the turbine is connected with the exhaust manifold. Turbocharger is modeled with its performance maps. Ricardo WAVE provides pre-processed standard compressor and turbine map files. Due to the unavailability of turbocharger map data for the actual system, these default map files provided by Ricardo wave are used in the present investigations. This may affect the accuracy of the simulation, however, good agreement between the results of simulation model and the experimental data for steady state shows that this affect is negligible on the accuracy of the simulation. The maps consist of 4 dimensional dataset: rotational speed, mass flow rate of the gas, pressure ratio and efficiency. The performance maps need pre-processing before they can be used by WAVE. There exists a routine in WAVE Build that has an algorithm to convert a performance map into a TCMAP object that WAVE uses to model both the compressor and the turbine [92].

In order to model the transient event, a controller is constructed. Three control elements are added to the model: a transient element, a switch and an actuator. Based on the transient event, the profile of the transient element can be set.

3.3 Modelling of techniques

This section provides details of the modeling strategies for torque assist, inertia reduction and air injection method. As stated earlier, the first two techniques are modelled using Lotus engine simulation and the third technique is modelled using Ricardo wave simulation software.

3.3.1 Modelling of torque assist and inertia reduction

A control system for providing the additional power to the turbocharger is constructed using sensors and actuators. The sensors in LES give the compressor power and turbine power. To simulate electric turbocharger assist system, an actuator is used which adds an additional torque to the turbocharger shaft. The angular velocity of the shaft is calculated using Newton's second law for rotational systems as given in the below equation [94]:

$$\dot{w} = \frac{1}{J_{tc}}(T_{qt} + T_{qja} - T_{qc})$$
(3.15)

where

 T_{qt} = Torque generated by turbine (Nm) T_{qja} = Torque from Jet-Assist (Nm) T_{qc} = Torque consumed by compressor (Nm) J_{tc} = Inertia of turbo system (kgm²)

The effect of change in inertia of compressor wheel on engine performance is investigated using different inertia values found as a percentage from the original compressor modelled with inertia as 6e-5 kgm². The reduction in inertia is then observed for 10%, 25% and 40%. The first two reductions are taken as motivation from Gilkes et al. [63]. As reported by [16], 50% inertia reduction brings faster engine response. Close to this value, the third reduction is chosen to observe the effect on the transient response of the engine.

3.3.2 Modelling of air injection

To regulate the air injection pressure, a control system is constructed using Simulink [95]. Ricardo wave software has limited control system capabilities when compared to other dedicated control packages such as Simulink. However, the wave package includes an interface to Simulink [96]. The control system in Simulink coupled with wave model is shown in Figure 3.4. Simulink accesses the wave signal through sensors and actuators. Signals from the sensors are send "Out of WAVE" whereas the signals to the actuators are sent "Into WAVE". In Simulink Mux block is used to create a vector of input signals while the Demux block is used to split the vector of output signals [92].

The air injection system is modeled by the addition of third ambient. Air is supplied through an extra ambient into the intake manifold. The mass flow of air is regulated by an orifice. Orifice models a restriction in the system such as to restrict the mass flow of the air. Its a joint style junction used to connect the two ducts. The orifice's diameter can be set equal to or smaller than the smaller of the two ducts [92]. The air with injection pressure P_1 is passed through an orifice for which the orifice flow (equation 3.16) gives the mass flow rate of the air [50].

$$\dot{m} = \frac{AP_1}{\sqrt{RT'}}\psi\left(\frac{P_1}{P_2}\right) \tag{3.16}$$

where ψ is defined as per the following equations:

For $P_2/P_1 > 0.528$,

$$\psi\left(\frac{P_1}{P_2}\right) = \left(\sqrt[\gamma]{\frac{P_2}{P_1}}\right) \sqrt{\frac{2\gamma}{\gamma - 1} \left(1 - \frac{\gamma}{\gamma - 1}\sqrt{\frac{P_2}{P_1}}\right)}$$
(3.17)

And for $P_2/P_1 \le 0.528$,

$$\psi\left(\frac{P_1}{P_2}\right) = \left(\frac{2\gamma}{\gamma - 1}\right) \left(\frac{\gamma + 1}{2(\gamma - 1)}\right) \sqrt{\gamma} \tag{3.18}$$

where

 P_1 : Pressure before the orifice

- P_2 : Pressure after the orifice
- A: Orifice area
- T': Gas temperature at the orifice inlet
- R: Gas constant
- γ : Ratio of specific heats

Orifice diameter can be changed in modeling through an actuator. In actual running case, an orifice flow meter can be used to monitor the flow rate of air and the flow of air can be controlled with a precision flow control valve [97]. A compressed gas tank can be utilized for providing the compressed air.

3. ENGINE SIMULATION



Figure 3.4: Wave coupled Simulink Model

Further, to observe the effect of air injection on emission characteristics of the engine, the parameters of the sensors to the Simulink model are altered. Following emissions are chosen at the sensors to investigate the effect of air injection:

- 1. CO emission
- 2. NO_x emission
- 3. HC emission

Ricardo wave simulation software enables to analyze the exhaust emissions. From the engine panel CO, HC and NO_x emissions are enabled under the combustion tab for the diesel engine.

3.4 CI engine test bed

The CI engine used to validate the simulation studies is available at the Advanced Automotive Laboratory of University of Huddersfield, UK. The schematic diagram of the experimental facility is shown in Figure 3.5 and the engine's characteristics are presented in Table 3.2. The test facility consists of a turbocharger, 220 kW AC dynamo-meter, pressure transducers, speed sensors, fuel flow meter and in-line torque

meter [98]. It is equipped with CADET software for programming the steady state and transient cycle. The test rig was utilized to conduct the steady state experiments. Before performing the experiment, the engine was run for 10 minutes to reach the steady state. Engine parameters (Brake power, Brake specific fuel consumption and Brake torque) were observed for the engine speed varying from 1000 - 2200 rpm. These steady state results provide useful data to validate the simulation based studies.



Figure 3.5: Schematic of experimental Facility [98]

Table 3.2:	Engine	block	specifications
			- P

Engine Type	4 Stroke Diesel	
Stroke	132 mm	
Bore	$103 \mathrm{~mm}$	
Number of Cylinders	4	
Combustion system	Direct injection	
Compression Ratio	18.3:1	
Connecting rod	144 mm	
Displacement	4.399 liter	

Table 3.3 shows the basic valve parameters of the engine. There are two valves per

cylinder for both intake and exhaust manifold. The diameter of intake valve is 36.2 mm and the exhaust valve is 33.5 mm.

Intake valves per cylinder	2
Intake valve diameter	$36.2 \mathrm{~mm}$
Exhaust valves per cylinder	2
Exhaust valve diameter	$33.5 \mathrm{~mm}$

Table 3.3: Basic valve parameters

3.5 Test procedures

In this section, two broad topics are covered: strategies for the transient events that the study covers and modelling of techniques investigated. The strategies that are applied in the present work are discussed which will provide a clear basis for carrying out the investigations under various operating conditions. Experimenting with different transient operating conditions is difficult as it requires expensive setups. Strategies are developed to model the techniques identified in literature review using the simulation codes mentioned in this chapter.

3.5.1 Strategies for transient events

In urban roads, the working condition of vehicle engine are mostly transient and constant speed and changing torque conditions are more popular [36] Hence, in the present study, more rigorous analysis is performed for load transients although acceleration transients are also investigated but to a lesser degree.

Furthermore, rapid increase in fueling rate also corresponds to a transient event as due to this elevated fueling and insufficient air flow in the combustion chamber, air fuel ratio becomes very low. Hence three types of transient events are considered for investigation: fueling rate change, speed transients and load transients. While comparing the technologies for relative improvement in transient response, fueling rate change is chosen to represent a transient event. After finding the optimum technology, detailed investigations are then performed for speed and load transient operating conditions.

3.5.1.1 Fueling rate change

The performance characteristics of the engine is affected by several parameters such as gas temperature and pressure in the cylinder, temperature and pressure characteristics of the turbocharger and the pressure variations during inlet and exhaust processes. The transient performance of the IC engine can be improved by supplying additional air into the cylinder and by controlling the fueling rate [97].

The transient performance of the engine is simulated through the rapid change in fueling rate which represents the transient event. Initially the model runs at 1000 rpm. During the transient, the quantity of fuel injected is increased from 40 mm³/injection to 52 mm³/injection and hence the boost pressure rises. The values of fueling rate have been referred from LES for modelling the transient response of turbocharged diesel engine [39].

3.5.1.2 Speed transients

Acceleration time and air injection pressure are the two important transient performance factors for which the turbocharger response is measured [97].

For the sake of studying the effect of acceleration time and injection pressure three tests are carried out as shown in Table 3.4. Test 1 presents the case of no air injection and only the acceleration time is varied by 1 sec and 2 sec. Test 2 is taken as the case of more rapid acceleration with the speed change in 1 sec and the injection pressure is varied. Finally, test 3 highlights the comparative improvement in turbo lag reduction for 1 sec and 2 sec injection duration respectively for the injection pressure obtained from test 2.

	Tosts	Turbocharger response		
	16515	Acceleration time	Air injection pressure	
	Test1	Varied	Not applied	
	Test2	Constant	Varied	
Test3 Varied		Varied	Constant	

Table 3.4: Test cases

3.5.1.3 Load transients

As discussed in section 2.1, constant speed and increasing torque (CSIT) is more popular than other load conditions. So, CSIT is the main focus for the investigations under load transient and these tests will be referred as "Load magnitude".

Nevertheless, narrow speed range transients are also demanding as discussed in the previous section. To this aim, investigations are also performed for constant torque and changing speed. These tests will be referred as "Constant load" and the results are

3. ENGINE SIMULATION

presented under section 5.4.1. Finally, scheduling of transient event is covered under section 5.4.3 to observe the effect of faster or slower load application on key performance indicators (KPI), the test will be referred as "Load schedule". For speed transients at a constant load step (section 5.4.1) and for load magnitude variations (section 5.4.2), transient time is taken as 1 sec which represents the case of rapid transient as discussed in section 3.5.1.2. For load scheduling, transient time is taken as 2 sec to observe the effect of slower load application (section 5.4.3). The tests described above for the three different load strategies are presented in Table 3.5.

Rather than randomly choosing the load steps, identification of load step is done based on engine performance degradation. It would be a novel approach of selecting the loads for the present investigation. Figure 3.6 shows the methodology for choosing the load steps. Different load step values are investigated for engine torque which is an important performance measure to observe the effect of various load steps. By the term torque deficit, it is meant the maximum percentage deviation of torque from its steady state value. Air injection is then simulated for the identified load steps and optimization of injection pressure is done based on the mass flow of the air passing through the orifice. The effect of optimized air injection is evaluated on turbocharger response parameters and performance improvement of the diesel engine under the transient events of load change is reported.

Table 3.5: Tests cases for load strategies

Tests	Load steps	Speed	Acceleration time
Constant load	Constant	Varied	Constant
Load magnitude	Identification based on torque deficit	Constant	Constant
Load schedule	Identification based on torque deficit	Constant	Varied

3.5.1.3.1 Constant load

Optimum injection pressure is considered based on the acceleration tests (to be carried out in Chapter 5) and the effect of this air injection on the transient response of the engine is evaluated for a load change from 80% to 100% in 1 sec at different speeds. This represents the case of higher engine load at a particular speed. Within the working range of the engine, five different engine speeds 1000 rpm, 1200 rpm, 1400 rpm, 1600 rpm and 1800 rpm are considered and the turbocharger response is observed for load change process at each speed.



Figure 3.6: Identification of load steps in load tests

3.5.1.3.2 Load magnitude

For final speed drop, the magnitude of applied load plays the prime role. In the initial steady state condition, the engine and the load torques are equal. With the application of new load, torque deficit occurs, when the engine torque can not instantly match with the increasing load torque [99]. The result is the engine speed drop which will be shown in the investigations.

To identify the load steps as discussed in section 3.5.1.3, initial load is taken as 10% and the final load is varied from 20% to 80%. As shown in figure 3.7, the effect of load change is more prominent for 70% to 76% final load. During the early load steps

3. ENGINE SIMULATION

(10% to 66%) engine torque continues to increase with time which indicates that during these load changes turbocharger is able to produce the required boost and the engine is capable to accept these loads. However, engine torque deteriorates during 70% to 76% final load and the engine speed becomes zero if the final load is further increased.



Figure 3.7: Variation of engine torque with different load steps in 1 sec, initial load 10%



Figure 3.8: Variation of brake torque with different load steps in 1 sec, initial load 50%

Similarly, decrease in torque is also observed for initial load 50% as shown in Figure 3.8. Torque deteriorates for 68% to 70% final load and engine speed becomes zero

if the final load is increased beyond 70%. The load steps with torque deficit are identified as load tests and are presented in Table 3.6. From the table it is clear that there are three load steps with 10% initial load and two with 50% initial load that need investigation for load acceptance.

Load Test	Initial load	Final load	Torque deficit
L1	10%	70%	1.5%
L2	10%	72%	2.7%
L3	10%	74%	4.0%
L4	10%	76%	6.0%
L5	50%	68%	0.4%
L6	50%	70%	4.5%

Table 3.6: Load tests resulting in torque deficit

3.5.1.3.3 Load schedule



Figure 3.9: Variation of brake torque with different load steps in 2 sec, initial load 10%

In load magnitude variation, load step is changed in 1 sec. In the present test, load step is varied in 2 sec to observe the effect of slower load application. As shown in Figure 3.9

3. ENGINE SIMULATION

and Figure 3.10, various initial and final load are investigated to find the decrease in torque under different load steps, in the same way as done in section 3.5.1.3.2. It is found that 10% to 68-76% and 50% to 68-74% load step corresponds to the decrease in torque which can be investigated under air injection for improvement.



Figure 3.10: Variation of brake torque with different load steps in 2 sec, initial load 50%

However, to observe the effect of load schedule, that load step is chosen that is common to both 1 sec and 2 sec and which brings maximum turbo lag reduction in 1 sec. So, the effect of air injection under that load step in 2 sec will be investigated in chapter 5 and the results will be compared with that of 1 sec.

3.6 Measurement uncertainty

The measured parameters generally show some dispersion from the mean values during engine testing. This is defined in terms of measurement error which shows the variation between measurements of same quantity on the same test. To obtain measurement error the same test needs to be repeated many times [100]. According to [101], precision index can be written as

$$S = \sqrt{\frac{\sum_{i=1}^{N} (X_i - \bar{X})^2}{N - 1}}$$
(3.19)

To report the uncertainty for the average value of N measurements, standard error is used which can be written as

$$\sigma_{\rm i} = \frac{S}{\sqrt{N}} \tag{3.20}$$

where S is the precision index as defined in equation 3.19 and σ_i is the standard error. The measurement repeated N times can then be typically written as [102]

$$X = \bar{X} \pm \sigma_{\rm i} \tag{3.21}$$

where \bar{X} is the mean of the dataset and σ_i is the standard error as defined in equation 3.20.

For the estimation of measurement uncertainty, the tests were repeated 5 times and the mean values were calculated for analysis. Brake torque, brake power and brake specific fuel consumption were the parameters that were investigated for accuracy. The standard error in this study is represented by \pm symbol as shown in Table 5.1, chapter 5.

3.7 Summary of the chapter

The focus of the chapter was on the simulation software used to meet the overall objectives of the study. Simulation models were developed and engine test bed was described for validation studies of the simulation model which will be covered in Chapter 5. Test procedures for implementing the various technologies of improving the transient response of the turbocharged diesel engine have been developed. Simulation strategies and various transient operating conditions were discussed in detail under the test procedures. Control systems were constructed for modelling the techniques.

3. ENGINE SIMULATION

$\mathbf{4}$

ELECTRIC TORQUE ASSISTANCE AND REDUCTION IN INERTIA OF COMPRESSOR WHEEL

Torque assistance method and reduction of inertia of compressor wheel are highlighted in the literature review as turbocharger specific measures to improve quality of transient response of turbocharged diesel engine. To meet objectives 1 and 2, these two methods are simulated in the present chapter and parametric study has been undertaken to analyze the quality indicators such as outlet pressure of the compressor and the compressor speed. The turbo lag is quantified to obtain the close to optimal transient response of turbocharged diesel engine. The Lotus Engine Simulation (LES) software is utilised which can predict full and part load performance under steady state and transient operating conditions.

The block diagram for the methodology is shown in Figure 4.1. It highlights the steps involved in the construction of simulation model as well as the control system for the improvement of transient response of the turbocharged diesel engine. The figure shows the system design as the initial step for simulation based on engine specification and operating parameters. A simulation strategy is then adopted to model the transient event. Turbo lag quantification is done for both torque assistance and inertia reduction method and based on turbo lag reduction, the transient response is reported.
4. ELECTRIC TORQUE ASSISTANCE AND REDUCTION IN INERTIA OF COMPRESSOR WHEEL



Figure 4.1: Block diagram for the proposed system

The length of the transient is taken as 6 seconds. The parameters for the quantification of turbo lag are listed in Table 4.1. To observe the output response of the turbocharger, compressor outlet pressure is taken as the major parameter for investigation under transient conditions. In addition, to analyse the turbocharger performance the effect of torque assistance is also investigated on turbocharger speed. To further analyze the transient period and to obtain quantitative estimates the rate of change of the parameters have been investigated. The quality factors listed in Table 4.1 are selected as transient response measure of the turbocharged engine.

S.No	Quality factors
1	Compressor exit pressure
2	Rate of change of compressor exit pressure
3	Compressor speed
4	Rate of change of compressor speed

 Table 4.1: Quality factors for transient test

4.1 Results and discussion

To meet the objectives for torque assistance and inertia reduction method, the simulation results of the proposed system are presented here. The modeling of the real system is considered by choosing the parameters from the literature. The simulations are performed using LES software for engine performance parameters as well as various turbocharging parameters. Since LES is well utilised by world wide clients and validated thoroughly at LOTUS [39], the simulation model in the present study is validated against the "Multi-cylinder turbocharged model" of Lotus engine simulation.

4.1.1 Steady state simulations

To study the engine performance during steady state operation, the variation of the engine parameters are presented in Figure 4.2 - 4.4. These figures show the engine performance characteristics with variation in engine parameters as a function of engine speed. The plots show the variation in BSFC (brake specific fuel consumption), brake torque, brake power and BMEP (brake mean effective pressure) which are the important engine performance indicators.

Figure 4.2 shows the torque rises rapidly to a maximum of around 178 Nm at 3000 rpm and then gradually decreases with the increase in engine speed. The brake mean effective pressure increases up to 11 bar at 3000 rpm then gradually decreases with engine speed as shown in Figure 4.3 At lower speeds i.e. below 3000 rpm BMEP drops off due to heat losses. While at higher speeds (above 3000 rpm) BMEP decreases because it becomes difficult to ingest a full charge of air.

4. ELECTRIC TORQUE ASSISTANCE AND REDUCTION IN INERTIA OF COMPRESSOR WHEEL



Figure 4.2: Brake torque



Figure 4.3: Brake mean effective pressure



Figure 4.4: Brake specific fuel consumption



Figure 4.5: Brake power

4. ELECTRIC TORQUE ASSISTANCE AND REDUCTION IN INERTIA OF COMPRESSOR WHEEL

BSFC reflects the efficiency of a combustion engine to produce power by burning the fuel. Figure 4.4 shows that the lowest value of specific fuel consumption is around 2000 rpm. At lower engine speed the fuel consumption increases due to increased time for heat losses from the gas to cylinder and piston wall. As the speed increases the fuel consumption also increases and hence the combustion efficiency decreases. The maximum power produced by the engine is about 64 kW at around 4000 rpm. Figure 4.5 shows the variation of Brake power with respect to engine speed. Initially it increases and then after 4000 rpm it starts decreasing.

As compared to steady state operation, intake air delay occurs during transient operation. Due to the deterioration of combustion boundary conditions during transient operations, the steady state operation findings cannot be applied directly and completely to transient operation. Nevertheless, transient operation studies should be based on steady state operation [103]. The steady state results presented in this section are comparable with the literature [39]. So, the information provided in this section related to steady state testing is useful. In the next section, emphasis is made on the transient operation as it is the event during which the engine performance deteriorates.

4.1.2 Transient simulation

In this section the results are presented for the effect of torque assistance and inertia reduction method under the event of fueling rate change as discussed in Chapter 3. Different values of torque assist are taken in the range of 0-0.32 Nm and investigations are performed to obtain the optimum value of torque assist. The value of the additional torque is motivated by the studies performed by Ibaraki et al. [61]. For this purpose, four different values are taken in the selected range as 0.08 Nm, 0.10 Nm, 0.16 Nm and 0.32 Nm. The response of compressor exit pressure for the different torque assist values is shown in Figure 4.6 The figure shows that initially by increasing torque assistance, the response is improved i.e.0.10 Nm torque assist offers more benefits than 0.08 Nm assist but further increase in torque to a value of 0.16 Nm gives almost similar but more stable response. Further increase in torque to a value of 0.32 Nm also gives almost same response. So the optimum value of torque assist is observed to be 0.16 Nm. This is consistent with the value reported in the literature [61].

Based on the above investigations 0.16 Nm torque is provided as an additional torque to the turbocharger shaft in the Lotus engine simulation environment for further evaluation. The Simulation is performed with and without electric torque assistance. The effect of torque assistance is noticeable in figure 4.7 Initially the compressor outlet pressure shows an initial delay in response before reaching the steady state. As shown in



Figure 4.6: Compressor exit pressure response for different value of torque assist

figure 4.7 that the compressor pressure reaches the steady state condition in 4.8 seconds with no torque assistance. The compressor outlet pressure response is significantly improved with torque assistance and the compressor pressure reaches the steady state in 1.8 second. The rate of change of compressor exit pressure response is plotted in figure 4.8. It shows much greater and faster rate of change for torque assist system as compared to without torque assist systems. The response time is reduced by 3 seconds with torque assist.

Figure 4.9 shows the effect of torque assistance on compressor speed response. When compared with no assist systems, it is clear that with torque assist the response time is improved by 3 seconds. This difference is more significant when rate of change of compressor speed response is plotted as illustrated in figure 4.10. In case of torque assistance, greater and quicker response is observed and the difference in time to reach the equilibrium is found to be 3 seconds.



Figure 4.7: Compressor exit pressure with and without torque assist



Figure 4.8: Rate of change of Compressor outlet pressure response



Figure 4.9: Compressor speed response with torque assist and without torque assist



Figure 4.10: Rate of change of compressor speed response

	Torque Assist					
Indicator		No Assist	Torque Assist			
	TL	Indicated Value	TL	Indicated Value		
CEP	4.8	4.8 2.13		2.2		
RCEP	3.5	0.39	0.5	3.5		
CS	5.6	141350	1.8	145354		
RCS	1.0	8310	0.5	154200		

Table 4.2: Comparison of the results obtained by torque assist method. TL: Turbo lag

The quantitative analysis of torque assist method is presented in Table 4.2. The quality indicators are compared for the case of torque assist and no torque assist. The analysis shows that the turbo lag is significantly reduced by the application of torque assistance. The compressor exit pressure (CEP) reaches the steady state condition in 4.8 seconds with no torque assistance. The compressor outlet pressure response is significantly improved with torque assistance and the compressor pressure reaches the steady state in 1.8 second. The rate of change of compressor exit pressure (RCEP) response shows much greater and faster rate of change for torque assist system as compared to without torque assist systems. The response time is 5.6 seconds without torque assist while with the torque assist the response time is 1.8 seconds. So the response time is improved by 3.8 seconds. Rate of change of compressor speed shows faster response with torque assist system. Moreover the results also show improvement in the indicated value for all the indicators with torque assist.

As discussed in chapter 1 and 2, the transient response can also be improved by reducing the inertia of the compressor. For inertia reduction, the values under consideration are shown in Table 4.3. It shows the reduction in compressor inertia whereas the initial value is $6e-6 \text{ kgm}^2$ and the reductions under consideration are -10%, -25% and -40%.

 Table 4.3:
 Reduction in compressor inertia

Original	$6.0e-6 \text{ kgm}^2$
-10%	$5.4e-6 \text{ kgm}^2$
-25%	$4.5e-6 \text{ kgm}^2$
-40%	$2.4e-6 \text{ kgm}^2$



Figure 4.11: Compressor exit pressure with reduction in inertia values



Figure 4.12: Compressor Speed with reduction in inertia values

Figure 4.11 shows the outlet pressure response of a turbocharger with reduction in

inertia by 10%, 25% and 40%. The response of the turbocharger with inertia reduction is compared with that of original inertia. It is depicted from the figure that although similar response is observed for different inertia reduction, the delay in the response of the compressor exit pressure is reduced by inertia reduction. However the reduction of delay is seen to be very small. This is in accordance with the previous studies [63]. The analysis will be more clear when turbo lag reduction will be quantified in the coming paragraphs.

To further analyze the effect of reducing the inertia on the transient response, compressor speed is plotted in Figure 4.12 with different values of inertia as described above. From the figure, it can be depicted that reducing the turbocharger inertia by 10% increases the compressor speed which is in accordance with equation 1.1 discussed in chapter 1. However, increasing the magnitude of inertia reduction doesn't bring improvement in compressor speed, both in terms of turbo lag and indicated value.

 Table 4.4: Comparison of the results obtained by inertia reduction method. TL: Turbo

 lag

Inertia		CEP	CS	
Reduction	TL	Indicated Value	TL	Indicated Value
Original	1.20	2.13	4	131901
-10%	1.10	2.14	3.8	133090
-25%	1.10	2.15	3.8	129576
-40%	1.10	2.15	3.8	118482

Table 4.4 shows the quantification of turbo lag for compressor exit pressure and compressor speed for the reduction in inertia of the compressor wheel. From the table it is clear that the improvement brought by this method is seen to be very small as the turbo lag is reduced by only fraction of seconds (0.2 sec) for both compressor exit pressure (CEP) and compressor speed (CS). It is worth noting that there is no benefit of decreasing the inertia below -10%.

4.2 Summary on electric torque assist and reduction in inertia of compressor wheel

- 1. Turbo lag decreases for compressor exit pressure with torque assistance of 0.16 Nm and the response time is reduced by 3 seconds.
- 2. Turbo lag decreases for compressor speed with torque assistance of 0.16 Nm and

the response time is improved by 3.8 seconds.

- 3. Rate of change of compressor exit pressure as well as rate of change of compressor speed shows much greater and faster response for torque assist system as compared to without torque assist systems.
- 4. Improvement in transient response by reduction in inertia is marginal and -10% reduction is the optimum value.

Here we achieved objective 1 and 2 by simulating passive methods of improving the transient response using Lotus engine simulation software. Further studies will be performed in the next chapter to investigate the results of active method of air injection for the improvement of the transient response of turbocharged diesel engine.

4. ELECTRIC TORQUE ASSISTANCE AND REDUCTION IN INERTIA OF COMPRESSOR WHEEL

$\mathbf{5}$

Air injection strategies

As per the literature review performed in chapter 2, three methods are highlighted that improve the transient response of turbocharged diesel engine. In the previous chapter, inertia reduction and torque assist method were found to improve the transient response of the engine. The findings of literature review indicate that air injection into the intake manifold is a highly promising measure to mitigate the drawback of poor transient performance of turbocharged diesel engine. Investigations are carried out in this chapter to show the effect of air injection at intake manifold on the transient response of the engine using modelling. Ricardo wave engine model is designed with the engine variable and operating parameters.

Figure 5.1 shows the pictorial representation of the different strategies used in the present study to simulate the transient response of the diesel engine. To find the optimum technology, the results of the three methods are compared by choosing common parameters and same simulation strategy. The effect of selected technology is then simulated for acceleration and load tests.



Figure 5.1: Strategies to model the transient response of diesel engine

5.1 Detailed investigation on air injection systems

The simulation results of the proposed system are presented here. Real engine is simulated using Ricardo wave for steady state and transient conditions. Based on the experimental data for steady state, the Ricardo wave model is validated for engine performance parameters such as brake torque, brake specific fuel consumption and brake power. The wave model is then modified to include air injection system at intake manifold. The control system is formulated using the co-simulation of Ricardo wave with Simulink [104].

5.1.1 Validation with real engine

Steady state testing is a useful tool for cross checking the available measurements in addition to the verification of experimental setup. The test results for the steady state are collected from the real CI engine and these results are used to validate the Ricardo wave model. It is assumed that the 1D gas dynamic code used in Ricardo wave has sufficient prediction capabilities to yield the reliable results. The nominal working range of the engine is 1000 - 2200 rpm. Engine performance parameters are chosen for

validation of Ricardo wave model with the real data. These are:

- 1. Brake Torque (BT)
- 2. Brake specific fuel consumption (BSFC)
- 3. Brake power (BP)

The validation of different engine performance parameters as a function of engine speed is shown in Figure 5.2, 5.3 and 5.4. Figure 5.2 depicts that initially torque rises at low engine speeds and attains the maximum value of 476 Nm at 1200 rpm and then gradually decreases with the increase in engine speed. It can be clearly observed from the Figure that the predicted values of brake torque by Ricardo wave model are in good agreement with the experimental data.



Figure 5.2: Validation of simulated values of brake torque with experimental data

Calculated values of BSFC were also satisfactory as shown in Figure 5.3. The validation of Ricardo wave with experiment shows a good agreement between the sim-

5. AIR INJECTION STRATEGIES

ulation values and experimental data for BSFC. The increase in fuel consumption at lower engine speed is due to the increased time for heat losses from the gas to cylinder and piston wall. The lowest value of specific fuel consumption is 0.201 kg/kWhr at 1800 rpm. Engine speed plays an important role for the fuel consumption in the engine. As the speed increases, the fuel consumption also increases due to increase in friction power and hence the efficiency decreases.



Figure 5.3: Validation of simulated values of brake specific fuel consumption with experimental data

Figure 5.4 shows the variation of brake power with respect to engine speed. Good agreement is obtained for brake power in the simulated and measured values. It is observed that the brake power increases with an increase in engine speed. The maximum power produced by the engine is about 79.58 kW at around 1800 rpm.



Figure 5.4: Validation of simulated values of brake power with experimental data

 Table 5.1: Comparison of Ricardo wave results with experimental data for performance parameters

Engine	Brake power (kW)		Brake torque (Nm)		BSFC (kg/kWhr)	
speed (rpm)	Experiment Ricardo		Experiment	Ricardo	Experiment	Ricardo
900	34.72 ± 0.22	27.6	371.75 ± 1.89	360.1	0.22 ± 0.004	0.228
1200	60.92 ± 0.74	56.4	476.45 ± 1.33	471.1	0.20 ± 0.002	0.201
1500	74.97 ± 0.30	74.2	471.05 ± 1.68	475.3	0.20 ± 0.0	0.200
1800	81.25 ± 0.29	81.0	422.70 ± 1.17	425.2	0.20 ± 0.0005	0.201
2100	80.20 ± 0.20	76.8	397.72 ± 1.03	394.49	0.20 ± 0.001	0.201

The quantitative measure of the agreement between the experiment and the simulation is presented in Table 5.1. It can be noted that Ricardo wave satisfactorily reproduces the experimental data for all the chosen performance parameters. The average difference between the experimental and the simulation data for brake power, brake torque and BSFC is 6%, 2% and 1% respectively. As discussed in section 3.2, the difference can be ascribed to the usage of default TCMAP in the simulation.

Henceforth, the calibrated wave model is expected to be able to predict the transient behavior of the chosen heavy duty compression ignition (CI) engine since the Ricardo wave has sufficient prediction capabilities to yield the reliable results for transient simulations.

5.2 Air injection system and its comparison with other methods of turbo lag reduction

Torque assist method and reduction in inertia method is investigated in sub section 4.1.2 to achieve objective 1 and 2 respectively. In this section, same simulation strategy is opted as described in section 3.5.1.1 to investigate the effect of air injection on the transient response of turbocharged diesel engine, covering objective 3. To know the effect of injection pressure, Figure 5.5 and Figure 5.6 are plotted. These figures show that the engine performance factors such as brake torque and brake power have noticed significant improvement with injection pressure up to 2.5 bar. Increasing the injection pressure after 2.5 bar doesn't bring significant improvement in these engine performance factors. So the injection pressure of 1 bar, 2 bar and 2.5 bar are chosen to perform an analysis of the effect of air injection on the response characteristics of the turbocharger.



Figure 5.5: Variation of brake torque with injection pressure



Figure 5.6: Variation of brake power with injection pressure



Figure 5.7: Effect of air injection pressures on compressor exit pressure

5. AIR INJECTION STRATEGIES

Air injection with 1 bar, 2 bar and 2.5 bar injection pressure is then simulated and the results are presented in Figure 5.7 and Figure 5.8. The results show the effect of air injection on compressor exit pressure (CEP) and compressor speed (C_Speed) respectively which are the same parameters as chosen in torque assist and inertia reduction methods. As shown in Figure 5.7 and Figure 5.8, with the application of air injection, turbo lag is reduced significantly for both the parameters for 1 bar, 2 bar and 2.5 bar when compared with no air injection. 2.5 bar air injection brings maximum improvement in the transient response of the turbocharger.

A comparison is now made for the assessment of relative improvement in transient response brought by the three methods. A value of 0.16 Nm is used for torque assist and inertia is reduced by 10% as per the conclusion from the investigations carried out in chapter 4. For air injection, 2.5 bar pressure is considered as it brings maximum improvement in transient response. It is clear from Figure 5.9 that when none of these methods is applied, compressor exit pressure and compressor speed rises rapidly but takes significant time to reach to the steady state, yielding greater turbo lag.



Figure 5.8: Effect of air injection pressures on compressor speed



Figure 5.9: Comparison of compressor exit pressure response (top) and compressor speed response (bottom) with torque assist method, air injection method, reduction in inertia method and original (no method applied)

The maximum achievable improvement in the transient response by the three methods is evaluated based on compressor exit pressure and compressor speed. The resulting turbo lag for compressor exit pressure and compressor speed is quantified as presented in Table 5.2. With the methods applied for transient response improvement, turbo lag is significantly reduced. However, the air injection method shows to outperform the other two methods, yielding the minimum turbo lag. The next section includes the detailed investigations on the effect of air injection system with different transient performance factors.

Mathad	Turbo lag (sec)		
Method	CEP	C_Speed	
Original	5.1	5.5	
Inertia reduction	2.2	4.9	
Torque assist	1.5	1.9	
Air injection	1.2	1.6	

Table 5.2: Comparison of turbo lag observed for compressor exit pressure (CEP) and compressor speed (C_Speed) response by different methods

5.3 Evaluation of air injection system under acceleration test

Acceleration tests represents the transient even at low engine speed and sudden acceleration. Simulations are performed for the acceleration tests to evaluate the effect of air injection systems on the transient response of the turbocharged diesel engine. Acceleration time and air injection pressure are varied under three tests as discussed in section 3.5.1.2 to observe the turbocharger response with the application of air injection at intake manifold. Air injection is simulated as discussed in section 3.3.2. These investigations will cover objective 5.

From Figure 5.2, it can be noted that the torque performance of the turbocharged diesel engine is poor at 1000 rpm engine speed. For these reasons simulations are performed at 1000 rpm engine speed with and without air injection during the transient conditions of rapid acceleration. During this time the engine speed is increased from 1000 rpm to 1700 rpm.

The time to acceleration is taken as 1 sec and 2 sec as they correspond to the case of rapid acceleration. For air injection, as discussed in section 5.2, 1 bar, 2 bar and

2.5 bar injection pressure are considered. To analyze the effect of varying the injection pressure, the simulation of air injection is performed to establish its effect first on the engine performance factors.

The major parameters investigated under acceleration tests are the following turbocharger response parameters:

- Compressor speed (CS) The speed at which the compressor is capable for continuous operation is known as compressor speed.
- Compressor exit pressure (CEP) The outward pressure available at the exit of the compressor is known as compressor exit pressure.
- Turbine inlet pressure (TIP) The exhaust gas pressure available at the exit of the engine cylinder which impinges into the turbine blade is known as turbine inlet pressure.
- Turbine inlet temperature (TIT) It is the temperature of the exhaust gases that enter the turbine unit.

These parameters indicate the output response of the turbocharger. So these are taken as a measure of transient response of the turbocharged engine.

5.3.1 Test 1

In the first test, the turbocharger response parameters are observed for the acceleration times of 1 sec and 2 sec for the speed change from 1000 rpm to 1700 rpm respectively. The effect of acceleration time on the response parameters of the turbocharger such as compressor exit pressure (CEP), turbine inlet pressure (TIP) and turbine inlet temperature (TIT) is depicted in Figure 5.10, 5.11 and 5.12 respectively. Similar trends are observed for turbo lag for all the chosen parameters of the turbocharger. For 1 sec acceleration time, the turbo lag values are found to be 6.4 sec, 6.8 sec and 8.0 sec for CEP, TIP and TIT respectively. On the other hand, the turbo lag values are 6.0 sec, 6.0 sec and 6.8 sec for CEP, TIP and TIT respectively for 2 sec acceleration time.



Figure 5.10: Effect of acceleration time on compressor exit pressure (CEP) response



Figure 5.11: Effect of acceleration time on turbine inlet pressure (TIP) response



Figure 5.12: Effect of acceleration time on turbine inlet temperature (TIT) response



Figure 5.13: Variation of response parameters (CEP, TIP and TIT) with acceleration time

5. AIR INJECTION STRATEGIES

For comparative analysis, the values are presented in Figure 5.13. It is clear from the figure that a decrease in time duration for acceleration from 2 sec to 1 sec increases the turbo lag and hence 1 sec corresponds to more rapid acceleration. The finding is consistent with the fact that turbo lag arises due to delayed response of the turbocharger and the faster is the transient event, comparatively slower will be the response. Based on this finding, 1 sec time duration for acceleration will be considered as the time duration for rapid acceleration for test 2.

5.3.2 Test 2

As per the investigations carried out in Test1, it should be noted that 1 sec acceleration time corresponds to rapid acceleration for which the improvement in transient performance needs to be measured. So in this test: the time for acceleration is taken as 1 sec and the effect of air injection pressure is measured on turbocharger response parameters such as CEP, TIP and TIT. The cases of additional air injection are compared with that of no injection. In the present investigation, the injection pressures considered are 1 bar, 2 bar and 2.5 bar to observe the optimum improvement in transient response.



Figure 5.14: Effect of air injection pressures on compressor exit pressure response



Figure 5.15: Effect of air injection pressures on turbine inlet pressure response



Figure 5.16: Effect of air injection pressures on turbine inlet temperature response



Figure 5.17: Variation of response parameters with air injection pressures for turbo lag

When additional air is injected at intake manifold, more air is supplied to the cylinder that optimizes the combustion processes. The consequence is the increase in exhaust energy which in turn increases turbine inlet pressure. This causes the turbocharger to rotate faster yielding improved turbocharger response [48]. The effect of air injection is noticeable as shown in Figure 5.14, 5.15 and 5.16. Initially the parameters show an initial delay in response before reaching the steady state. The recovery time with air injection is significantly reduced as compared to that of no air injection for all the parameters. Turbo lag decreases and the final steady state values are also improved with the increase in air injection pressure. The comparative reduction in turbo lag with varied air injection pressure for each response parameter is shown in Figure 5.17.

Although each of the selected injection pressure is sufficient to improve the transient performance as shown in Figure 5.14, 5.15 and 5.16, however, it is important to find the optimum value of the injection pressure. So in addition to the performance improvement evaluation with air injection, it is worth calculating that how much energy is imparted with the air injection. Turbo lag reduction with minimum energy required by air injection will be an additional source of information for selecting the optimum injection pressure. The energy supplied per sec (E) through air injection can be calculated as:

$$E = \frac{P_1 \dot{m}}{\rho} \tag{5.1}$$

where P_1 is the pressure before the orifice, \dot{m} is the mass flow rate of the air passing through the orifice and ρ is the air density. Mass flow rate of the air passing through the orifice can be calculated from equation 3.16, described in chapter 3.



Figure 5.18: Dependency of turbo lag reduction on air supply parameters for compressor exit pressure (CEP)

Figure 5.18, 5.19 and 5.20 show the reliance of turbo lag reduction on air assist pressure and in turn the energy required for each of the selected performance parameter. Higher the injection pressure, more the energy required to significantly reduce the turbo lag. For comparative analysis, the turbo lag reduction per unit energy corresponding to the injection pressure is presented in Table 5.3. It can be concluded from the table that the maximum improvement in percent turbo lag reduction/energy is brought by air injection corresponding to 1 bar pressure. Hence, 1 bar air injection is the optimum injection pressure with percent turbo lag reduction per unit energy as 0.290 per joule for compressor exit pressure, 0.392 per joule turbine inlet pressure and 0.555 per joule for turbine inlet temperature (TIT). The optimum points are highlighted with red colour in the figures.



Figure 5.19: Dependency of turbo lag reduction on air supply parameters for turbine inlet pressure (TIP)



Figure 5.20: Dependency of turbo lag reduction on air supply parameters for turbine inlet temperature (TIT)

Injection	Turbo lag reduction per unit energy				
Pressure (bar)	CEP	TIP	TIT		
1	0.290	0.392	0.555		
2	0.102	0.122	0.146		
2.5	0.032	0.034	0.040		

Table 5.3: Turbo lag reduction per unit energy corresponding to air injection pressure for turbocharger response parameters (CEP, TIP and TIT)

5.3.3 Test 3

In this test 1 bar is taken as the optimum injection pressure as found in test 2. The response characteristics are observed for acceleration time of 1 sec and 2 sec respectively for the speed change from 1000 rpm to 1700 rpm as shown in Figure 5.21. As per the comparative analysis illustrated in the figure, the turbo lag quantification for the air injection of 1 bar highlights a remarkable reduction in turbo lag for 1 sec transient as compared to that of 2 sec. This difference is more perceptible when the rate of change of response parameters is plotted as shown in Figure 5.22, 5.23 and 5.24.



Figure 5.22: Rate of change of compressor exit pressure



Figure 5.21: Effect of air injection on response parameters with variable acceleration time



Figure 5.23: Rate of change of turbine inlet pressure



Figure 5.24: Rate of change of turbine inlet temperature

The figures show that faster response is observed for 1 sec when air injection is applied. Hence more improvement is achieved where the turbo lag is more prominent (that is for 1 sec acceleration time as reported in Test1). This ensures that the air injection at intake manifold greatly improves the transient response of turbocharged diesel engine for the event of rapid acceleration.

5.4 Evaluation of air injection system under load test

Load step demand of the engine is another important event that limits the transient response of the turbocharger. In urban roads, the engine speed and the load varies suddenly and frequently, resulting in increased exhaust emissions. In such operations, the effect of air injection technique to access the transient response of the engine is of great interest. To this aim, investigations under speed transient are already presented in the previous section and the effect on exhaust emissions will be covered in the next chapter. In this section, response of the turbocharged diesel engine is investigated for different load conditions. The method of analysis and the overall tests are summarized in Figure 5.25. Three cases of load transients are considered: constant load, load magnitude variation and load scheduling. Air injection technique is simulated and after

5. AIR INJECTION STRATEGIES

optimization of injection pressure, its effect on the transient response is presented.



Figure 5.25: Schematic of load tests

5.4.1 Turbocharger response for constant load step

The transient performance of the turbocharged diesel engine is investigated in terms of turbocharger response parameters. As discussed in the test procedures in Chapter 3, the load is kept constant with initial load 80% and the final load 100%. Five engine speeds are chosen within the operating range of the engine. Optimization performed in acceleration test (Test 2) for air injection is referred and based on that, the effect of 1 bar air injection is observed on turbocharger response parameters.

Results are presented in two types of diagrams. The first ones (Figure 5.26, Figure 5.27 and Figure 5.28) show the effect of 1 bar air injection on compressor exit pressure (CEP), turbine inlet pressure (TIP) and turbine inlet temperature (TIT) for engine speeds 1000 and 1800 rpm under the constant load to show the effect of air injection at lower and higher engine speed respectively. The second ones (Figure 5.29, Figure 5.30 and Figure 5.31) show the dependence of turbo lag reduction (%) and steady state value ratio on engine speed. By the term steady state value ratio (SS Value ratio), it is meant the ratio of the final steady state value of the response parameter for air injection to that of no air injection. In addition to turbo lag reduction, SS Value ratio is used as an indicator for the effect of air injection on the equilibrium state value of the response parameter.



Figure 5.26: Effect of air injection on compressor exit pressure response during a load change from 80% to 100% at different engine speeds



Figure 5.27: Effect of air injection on turbine inlet pressure response during a load change 80% to 100% at different engine speeds


Figure 5.28: Effect of air injection on turbine inlet temperature response during a load change 80% to 100% at different engine speeds

Figure 5.26, Figure 5.27 and Figure 5.28 compare the turbocharger response parameters with air injection and without air injection when the engine is subjected to a load step from 80% to 100% in 1 sec under constant engine speeds of 1000 and 1800 rpm. Figure 5.26 and Figure 5.27 show that as the speed increases CEP and TIP also increases that supplies more energy to the turbocharger. Figure 5.28 shows that the maximum attainable value of TIT is lower when air injection is applied at lower engine speed (1000 rpm). While at higher engine speed (1800 rpm) air injection brings same improvement in TIT also.

To further analyse the results obtained above, steady state value ratio of the response parameter under consideration and turbo lag reduction (%) are plotted with respect to the engine speed. Figure 5.29, Figure 5.30 and Figure 5.31 show the dependency of steady state value ratio of the response parameter under consideration and turbo lag reduction (%) on the engine speed. It can be seen from these figures that in terms of SS value ratio, the effect of air injection is more at higher engine speeds (1600-1800 rpm) as greater torque and power are produced while at lower engine speeds (1000-1400 rpm) the technique brings far less improvement.



Figure 5.29: Dependency of turbo lag and SS value ratio on engine speed for compressor exit pressure (CEP)



Figure 5.30: Dependency of turbo lag and SS value ratio on engine speed for turbine inlet pressure (TIP)

5. AIR INJECTION STRATEGIES



Figure 5.31: Dependency of turbo lag and SS value ratio on engine speed for turbine inlet temperature (TIT)



Figure 5.32: Dependency of transient response improvement on engine speed in terms of turbo lag reduction (%)

To access the effect of air injection on the transient performance of the diesel engine under these operating conditions of constant load and speed variation, Figure 5.32 is plotted. From the figure it can be seen that significant improvement in transient performance is observed for CEP, TIP and TIT at 1000-1400 rpm engine speed in terms of turbo lag reduction (%) while at 1600-1800 rpm engine speed, air injection yielded lesser improvement in the transient performance.

5.4.2 Load magnitude variations

Initial engine speed is 1000 rpm and load magnitude is varied in 1 sec. The key performance indicators (KPI) are chosen based on engine and turbocharger response. Various load steps of increasing magnitude are investigated and the effect is observed on these KPI:

- Engine speed
- Compressor exit pressure (CEP)
- Turbine inlet pressure (TIP)
- Turbine inlet temperature (TIT)

5.4.2.1 Effect of air injection for varying load magnitude

Identification of load step is an important aspect in the load transients to consider those load steps whose application can adversely affect the engine performance. As discussed in section 3.5.1.3.2, the results of L1 to L6 load tests are analysed and discussed for the effect of air injection. A series of demonstrative diagrams are provided to analyze the effect of changing the mass flow of the injected air on the injection pressure. This will provide optimization of air injection followed by the transient response improvement analysis for the effect of optimized air injection.

With the application of load step in 1 sec, the engine speed decreases for L1 to L4 load tests (Initial load 10%) as shown by "No Injection" curves in Figure 5.33. Higher the load magnitude, greater is the speed drop. To improve the engine response, air injection is applied at intake manifold. Minimum air injection pressure, required to prevent zero engine speed are 1.5 bar, 2.6 bar, 4.4 bar and 6 bar for L1, L2, L3 and L4 load tests respectively. However, it is noted from the figure that these injection pressure are not sufficient to prevent the engine from stalling which is observed for both, no injection as well as air injection for the given load tests.



Figure 5.33: Effect of air injection on engine speed with different load steps, initial load 10%



Figure 5.34: Effect of air injection on engine speed with different load steps, initial load 50%

Similarly, in case of 50% initial load, L5 and L6 load tests need investigation. When there is no air injection, the speed decreases as shown in Figure 5.34. With the application of air injection with minimum injection pressure, speed gradually increases but takes too long to reach the steady state. It takes 19 sec with 2.4 bar air injection for L5 load test and 18 sec with 4.2 bar air injection for L6 load test.

So, based on the engine speed response for air injection under two tests, it is worth considering higher injection pressure to obtain the equilibrium state of the engine speed for the load tests L1 to L6. But injection pressure can not be increased randomly as it may cause surging [99]. Optimization of injection pressure is hence essential to avoid this situation. In the next section, optimization of injection pressure is done based on the mass flow rate of the injected air which depends upon the orifice diameter as discussed in Chapter 3 (equation 3.16).

5.4.2.2 Optimization of injection pressure

In the investigations performed on speed transient, optimum injection pressure is obtained based on the energy imparted per second. From equation 3.16 and equation 5.1 it can be depicted that energy imparted depends upon the orifice diameter which regulates the mass flow of the injected air. So, rather than calculating the energy imparted per second for each load step, orifice diameter can be chosen as a good choice of parameter for the purpose of optimization of injection pressure bringing novelty in the optimization process.

From the different load steps for which the engine speed deteriorates, following load steps are considered to observe the effect of increasing the load magnitude:

- 1. Initial load 10%, final load increased from 70% to 76% (Load tests L1 and L4).
- 2. Initial load 50%, final load increased from 68% to 70% (Load tests L5 and L6).

For the optimization of air injection, L1 test is chosen and the results are presented in Figure 5.35. The figure shows the effect of varying the orifice diameter on the engine speed for load step L1. It is observed from the results that there is a correlation between the orifice diameter, injection pressure and turbo lag value.

Top left plot of Figure 5.35 for orifice diameter 5 mm shows the application of higher injection pressures to overcome stalling. It shows that higher the injection pressure, better the transient response in terms of turbo lag. Minimum required pressure to overcome zero speed is 1.8 bar and the turbo lag is 15 sec. Increasing the injection pressure decreases the turbo lag and 4 bar yields the minimum turbo lag of 13 sec. So,

5. AIR INJECTION STRATEGIES

for 5 mm orifice diameter, the optimum injection pressure is 4 bar with turbo lag 13 sec. Further, optimum injection pressure is observed for 10 mm, 15 mm and 20 mm orifice diameter. In the same manner, optimum injection pressure and the turbo lag value is noted for different orifice diameter for L4, L5 and L6 load tests. The quantification is presented in Table 5.4.



Figure 5.35: Engine speed with varying orifice diameter and injection pressures for load test L1

Table 5.4: Dependency of injection pressure (IP in bar) and turbo lag (TL in sec) on orifice diameter(mm) for the different load steps

	Load tests								
OD	L1		L4		L5		L6		
	IP	TL	IP	TL	IP	TL	IP	TL	
5	4	13	8	12	5	12	6	12	
10	3	10	4	10	4	10	3	10	
15	3	9	3	9	3	10	3	9	
20	3	10	3	10	4	11	3	10	

It can be noted from Table 5.4 that orifice diameter considerably affects the injection pressure and turbo lag. This can be explained from equation 3.16 which indicates that the mass flow rate of air is directly proportional to the area of the orifice. So increasing the orifice diameter increases the mass flow rate of air which in turn increases the air fuel ratio. This leads to better combustion and hence turbo lag is reduced.

This dependency of injection pressure and turbo lag on orifice diameter for each test is demonstrated in Figure 5.36 to Figure 5.39. For each test, it can be noted from the figures that at 5 mm orifice diameter, turbo lag is considerably high (13 sec for L1 and 12 sec for L4, L5 and L6 each). Also higher injection pressure is required (4 bar for L1, 8 bar for L4, 5 bar for L5 and 6 bar for L6) at 5 mm orifice diameter.



Figure 5.36: Correlation between turbo lag, injection pressure and orifice diameter for L1 test



Figure 5.37: Correlation between turbo lag, injection pressure and orifice diameter for L4 test

It can be analyzed from the figures that increasing the orifice diameter from 5 mm to 10 mm drastically reduces the injection pressure and turbo lag for the different load steps. Increasing the orifice diameter to 15 mm yields 3 bar injection pressure with minimum turbo lag. Nevertheless, increasing the orifice diameter beyond 15 mm doesn't bring significant improvement in turbo lag. A dip in turbo lag value at 15 mm orifice diameter for all the chosen tests (L1, L4, L5 and L6) indicates the peculiarity of these tests. This means that further increasing the mass flow rate of air by increasing the orifice diameter may cause the compressor to surge as discussed in section 5.4.2.1. Due to this reason turbo lag increases after 15 mm orifice diameter.



Figure 5.38: Correlation between turbo lag, injection pressure and orifice diameter for L5 test



Figure 5.39: Correlation between turbo lag, injection pressure and orifice diameter for L6 test

So, for the purpose of optimization, orifice diameter will be taken as 15 mm and the effect of 3 bar air injection for the different load steps will be investigated on other key performance parameters.

5.4.2.3 Effect of air injection on turbocharger response parameters

As concluded in the previous section, orifice diameter is taken as 15 mm and the effect of 3 bar air injection for the two increasing load steps (L1 to L4 and L5 to L6) will be investigated in this section on the turbocharger response parameters: CEP, TIP and TIT. As shown in Table 5.5, the TL reduction (%) is noted for CEP, TIP and TIT after the application of air injection.

Table 5.5: Turbo lag reduction (%) for the turbo charger response parameters under different load tests

Load	TL reduction $(\%)$							
step	CEP	TIP	TIT					
L1	33	50	45					
L4	27	26	25					
L5	25	42	42					
L6	45	50	45					



Figure 5.40: Effect of air injection on turbocharger response parameters for different load magnitude

Figure 5.40 shows that air injection significantly reduces the turbo lag for the turbocharger response parameters under the effect of different load steps. Nevertheless, as per the quantification presented in Table 5.5, increasing the magnitude of stronger load (Initial load 10%) decreases the turbo lag reduction. This is because of the fact that it is difficult for the engine to cope with high loading instantly [16]. Whereas, increasing the magnitude of lighter load (Initial load 50%) increases the turbo lag reduction. Maximum improvement in turbo lag reduction is observed for 50-70% load step (Load test L6).

5.4.3 Load schedule

Based on the results of the previous section, air injection under Test L6 brings maximum turbo lag reduction in 1 sec. So, for the purpose of observing the effect of load scheduling, the effect of air injection for the same load step (Test L6) in 2 sec will be investigated in the present section and the results will be compared with that of 1 sec.



Figure 5.41: Engine speed with varying orifice diameter and injection pressures for load step 50-70% in 2 sec

Before modeling the effect of air injection, again optimization of injection pressure

5. AIR INJECTION STRATEGIES

based on orifice diameter is done as shown in Figure 5.41. From the analysis of the figure, optimum value of air injection is found to be 3 bar at 15 mm orifice diameter. So, the effect of 3 bar air injection at 15 mm orifice diameter is investigated on turbocharger response parameters.



Figure 5.42: Effect of air injection on turbocharger response parameters for different load schedules for L6 load test

Figure 5.42 shows the effect of air injection on CEP, TIP and TIT under L6 load test for 1 sec and 2 sec load schedules. Significant reduction in turbo lag is observed for these three parameters, however, the turbo lag reduction (%) is more for 1 sec than for 2 sec. So, air injection technique is more effective for 1 sec load transient than for 2 sec. This behaviour is similar to that of acceleration tests summarizing that the air injection technique is more effective for 1 sec transients as compared to 2 sec.

5.5 Summary of the chapter

In this chapter, real engine is modeled using Ricardo wave simulation software and after the validation using the steady state simulations, the transient performance of the turbocharged diesel engine is investigated. Air injection at intake manifold is simulated to improve the transient response. The transient performance of the diesel engine is investigated under various acceleration tests. Finally, the effect of air injection on transient response of the turbocharged diesel engine is simulated under different load conditions. Based on the results obtained from this study the following conclusions are drawn for this specific heavy duty diesel engine:

1. Engine performance parameters are tested for steady state and Ricardo wave successfully reproduces the experimental data for the chosen performance parameters

i.e. brake torque, brake specific fuel consumption and brake power.

- 2. Comparison is made between the three methods of transient response improvement and of the three methods considered, air injection is identified as the most efficient technique with respect to turbo lag reduction.
- 3. The turbocharger response parameters such as compressor exit pressure (CEP), turbine inlet pressure (TIP) and turbine inlet temperature (TIT) increases with the increase in air injection pressure and 2.5 bar injection pressure brings the maximum improvement in the transient response.
- 4. The amount of energy imparted for air injection should not be neglected while reducing the turbo lag for the improvement in transient response. Greater reduction in turbo lag can be achieved by applying higher assist pressure up to 2.5 bar but at the cost of compromised energy imparted per second.
- 5. 1 bar is the optimum injection pressure with turbo lag reduction per unit energy as 0.290, 0.392 and 0.555 per joule for compressor exit pressure, turbine inlet pressure and turbine inlet temperature (TIT) respectively.
- 6. Turbo lag is more prominent when the acceleration time is 1 sec than that for 2 sec. With the application of air injection, faster recovery time is observed for 1 sec rapid acceleration than that of 2 sec.
- 7. Under constant load step, both the turbo lag reduction as well as the value of the maximum attainable parameter depends upon the engine speed. Air injection brings maximum improvement in turbo lag reduction at lower speed range (1000-1400 rpm) while its effect on maximum attainable value is observed at higher speed range (1600-1800 rpm).
- 8. 3 bar air injection at 15 mm orifice diameter is found to be the optimum value of air injection and its effect on turbocharger response is investigated under load magnitude variations. The results show that with air injection, although the turbo lag reduces under all the load steps, the improvement is more noticeable for lighter load variation than for stronger load. Maximum improvement under load magnitude increment is observed for 50-70% load step.
- 9. The effect of changing the load schedule from 1 sec to 2 sec is observed on turbocharger response parameters. Optimization of injection pressure shows 3 bar air injection at 15 mm orifice diameter to be more effective for 1 sec load schedule than for 2 sec indicating more benefit in case of faster load application.

6

EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION

Today, environment protection has become a topic of central concern. Adverse effects of diesel fuel on human health and environment have led the governments to put forward the regulations for permissible exhaust emissions. It is expected that emission legislation will be further highlighted to achieve 'near zero' emissions by 2030 [105]. Euro standards are developed by Europe lowering the emissions requirement continuously from Euro I to Euro VI since 1993. Euro VI legislation demands that CO, HC and NO_x emissions should not exceed 1.5 g/kWh, 0.13 g/kWh and 0.4 g/kWh respectively [66]. Present chapter deals with the effect of air injection technique on exhaust emissions and to ensure that the application of the technology comply with Euro VI standard.

The effect of air injection on the transient response of the turbocharged diesel engine is investigated in Chapter 5. The results of the investigation show that with the application of air injection at intake manifold, the transient performance of the turbocharged diesel engine is improved under the transient events of fueling rate change, speed transient and load transient. This indicates that the delay between fueling and air-supply is reduced by the application of air injection. The consequence would be

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION

better air supply to the cylinders that improves combustion. The exhaust emissions that are generated due to poor combustion following the turbo-lag period should be reduced. To this aim, investigations are carried out in the present chapter to analyze the effect of air injection on exhaust emissions.

For the overall optimization of the system, it is important to ensure that the air injection technique doesn't increase the exhaust emissions. For this purpose, optimization of injection pressure and mass flow of the air through the orifice diameter is performed and exhaust emissions are analysed. This will provide novel contribution as firstly the effect of air injection technique is rarely evaluated for exhaust emissions and secondly optimization of air injection for satisfying Euro 6 standard will be a new and important study due to emission requirement legislations. This will pave way for further research in this area. In the present chapter, investigations are performed to achieve objective 8 and objective 9.

6.1 Emission characteristics under speed transient

As discussed in chapter 5, low speed transient - 1000 rpm to 1700 rpm in 1 sec corresponds to the case of rapid acceleration and air injection is found to improve the transient performance of the engine under this case. To evaluate the effect of air injection on exhaust emissions and to comply with Euro 6 standard, optimization of injection pressure based on mass flow of the air through the orifice diameter is done for the speed transient of 1000 - 1700 rpm in 1 sec. Explicit multiple diagrams are provided to describe the effect of orifice diameter and in turn injection pressure on emissions.

6.1.1 Optimization of air injection for emissions under speed transient

Air injection pressure is varied and its effect on exhaust emissions (CO, HC and NO_2) is observed for 5 mm, 10 mm and 15 mm orifice diameter. The results are presented in Figure 6.1, Figure 6.2 and Figure 6.3.

The variation of CO emission values with injection pressure and orifice diameter is shown in Figure 6.1. It is clear from the figure that in case of CO emissions, 1 bar injection pressure increases the emissions if compared with 2 bar and 3 bar. The emissions also increase with the increase in orifice diameter. 2 bar injection pressure reduces the CO emissions for 5 mm and 10 mm orifice diameters. Further increasing the orifice diameter to 15 mm increases the CO emissions for 2 bar also. Whereas, 3 bar air injection significantly reduces CO emission particularly for 10 mm. There is no benefit of further increasing the injection pressure or orifice diameter and hence 3 bar



injection pressure at 10 mm orifice diameter is the optimum value for CO emissions.

Figure 6.1: CO emissions with varying injection pressures and orifice diameter



Figure 6.2: HC emissions with varying injection pressures and orifice diameter

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION

Figure 6.2 depicts the HC emissions with varying injection pressure and orifice diameter. The analysis of the figure shows similar trend for CO emissions. 1 bar injection pressure is not sufficient to reduce the HC emissions for 5mm, 10 mm and 15 mm orifice diameter. Increasing the injection pressure up to 3 bar decreases the emissions and minimum HC emissions are observed for 3 bar air injection at 10 mm orifice diameter.

Significant reduction in NO_2 emissions is observed with air injection as shown in Figure 6.3. Increasing the injection pressure doesn't make much difference in the relative improvement in NO_2 emission reduction and hence 1 bar injection pressure at 10 mm is enough to minimize the emission as shown in Figure 6.3.



Figure 6.3: NO_2 emissions with varying injection pressures and orifice diameter

Diesel engine emissions typically contains NO and only 5% is the NO₂ [92], so for the purpose of evaluating the emissions against Euro 6 standard, only CO and HC emissions will be considered. Figure 6.4 is plotted for CO and HC emissions for 3 bar air injection and varying orifice diameter. With 3 bar air injection at 10 mm orifice diameter, CO and HC emissions are 1.5 g/kWh and 0.107 g/kWh respectively which satisfies Euro 6 emission standard [66] as shown with green color bar in Figure 6.4. So, the optimum injection pressure for CO and HC emission is 3 bar at 10 mm orifice diameter. After

finalizing the optimum injection pressure, its effect on the concentration of CO, HC and NO_x emissions is evaluated in the next section.



Figure 6.4: Optimization of air injection for CO and HC emissions at 1700 rpm engine speed

6.1.2 Emission reduction under the effect of air injection for speed transient

In this section, the effect of 3 bar air injection on emissions for 10 mm orifice diameter is compared with no air injection and the percent reduction is quantified for comparison. The concentration of the emissions are noted in parts per million (ppm). CO, HC and NO_x emissions are plotted as shown in Figure 6.5. It can be concluded from the figure that the air injection doesn't increase the emissions. Slight reduction is observed for CO (5%) and HC (0.4%) emissions while significant reduction (75%) is observed on NO_x emissions.

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION



Figure 6.5: Effect of air injection on emission concentration under speed transient: CO emission (left), HC emission (right) and NO_x emission (bottom)

6.2 Emission characteristics under load transient

Effectiveness of air injection technique under load transient is investigated in chapter 5. 3 bar air injection at 15 mm orifice diameter is found to be most effective under 50-70% load step in 1 sec for improving the turbocharger response. In the present section, emission characteristics of the turbocharged diesel engine are investigated under 50-70% load step in 1 sec. CO, HC and NO₂ emissions are analyzed. Optimization of injection pressure and orifice diameter is done for these emissions.

6.2.1 Optimization of air injection for emissions under load transient

Air injection pressure and orifice diameter are varied and CO, HC and NO₂ emissions are plotted as shown in Figure 6.6, Figure 6.7 and Figure 6.8. Figure 6.6 shows that increasing the injection pressure increases CO emissions. For 1 bar air injection, CO emissions are minimum for all the chosen orifice diameters. Minimum value of CO emissions are observed for 15 mm orifice diameter with 1 bar air injection. Figure 6.7 and Figure 6.8 show that for HC and NO_2 emissions, 3 bar air injection at 10 mm orifice diameter and 2 bar air injection at 15 mm orifice diameter respectively yields minimum emissions.



Figure 6.6: CO emissions with varying injection pressures and orifice diameter

However, it is noted that in case of load transient, emissions are below Euro 6 standard for all the chosen injection pressures and orifice diameters. Hence, optimization is done based on the injection pressure and orifice diameter for which all the emissions with air injection are less than no air injection. For this purpose, quantification of emissions are done for each pressure and orifice diameter as presented in Table 6.1, Table 6.2 and Table 6.3. In order to get reduction in all the three emissions when applying air injection, 1.2 bar air injection at 10 mm needs to be considered as it brings reduction in all the three emissions.

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION



Figure 6.7: HC emissions with varying injection pressures and orifice diameter



Figure 6.8: NO_2 emissions with varying injection pressures and orifice diameter

OD	Injection pressure							
(mm)	No Injection 1 bar 1.2 bar 2 bar 3 bar							
5	0.76	0.71	0.74	0.79	0.82			
10	0.76	0.64	0.757	0.91	0.98			
15	0.764	0.491	0.7642	1.16	1.92			

Table 6.1: Dependency of CO emissions on injection pressure and on orifice diameter

Table 6.2: Dependency of HC emissions on injection pressure and on orifice diameter

OD	Injection pressure								
(mm)	No Injection 1 bar 1.2 bar 2 bar 3 bar								
5	0.104	0.1049	0.1046	0.1041	0.1046				
10	0.104	0.105	0.1039	0.1028	0.1022				
15	0.104	0.1074	0.1046	0.1023	0.1066				

Table 6.3: Dependency of NO₂ emissions on injection pressure and on orifice diameter

OD	Injection pressure								
(mm)	No Injection 1 bar 1.2 bar 2 bar 3 bar								
5	18.15	18.21	18.11	17.93	17.84				
10	18.15	18.42	18.08	17.41	17.12				
15	18.15	18.99	18.05	17.14	18.82				

6.2.2 Emission reduction under the effect of air injection for load transient

The effect of 1.2 bar air injection at 10 mm on the concentration of emissions under load step 50-70% in 1 sec is then compared with no air injection and the percent reduction is quantified. As shown in Figure 6.9, CO, HC and NO_x emissions are observed in ppm. Percent reduction by the application of air injection is 0.8%, 0.01% and 0.4% for CO, HC and NO_x respectively as shown in Figure 6.9. So, the emissions doesn't increase with the application of air injection under the event of load change also.

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION





6.3 Overall optimization of air injection system

The effect of air injection on the transient performance of the diesel engine for different types of transient operating conditions is presented and discussed in the previous chapters. For the overall optimization of the system, it is important to consider the environmental performance in addition to the engine performance improvement. The present chapter presented the effect of air injection on exhaust emissions.

Table	6.4:	Quantification	of the	effect	of air	injection	in terms	of	energy	imparted	per
second.	P: ir	ijection pressur	e (bar)	, OD:	Orifice	e diameter	r (mm)				

Monsuro	Speed tran	isient	Load transient			
Measure	Air injection	Energy	Air injection	Energy		
Performance Characteristics	P: 1 bar OD: 5 mm	99 J	P: 3 bar OD:15 mm	2.5 kJ		
Emission Characteristics	P: 3 bar OD: 10 mm	2.7 kJ	P: 1.2 bar OD: 10 mm	446 J		

Table 6.4 quantifies the effect of optimum value of air injection on performance and

emission characteristics under speed and load transient operating conditions. For practical considerations, its worth calculating the mass flow rate of the air corresponding to optimum air injection injection reported in chapter 5 and 6 and hence the energy imparted per second. From the table it can be depicted that optimum air injection is 1 bar at 5 mm orifice diameter and 1.2 bar at 10 mm for speed and load transient respectively as these injections correspond to minimum energy imparted per sec.

6.4 Summary of the chapter

The exhaust emissions CO, HC and NO_x are analysed for the effect of air injection technique under speed and load transients. Following conclusions are drawn from the investigations:

- 1. Optimum injection pressure is 3 bar at 10 mm orifice diameter that satisfies Euro 6 standards for CO and HC emissions under speed transient.
- 2. Air injection technique under speed transient doesn't increases the concentration of exhaust emissions. Rather, 5% and 0.4% reduction is observed in the concentration of CO and HC emissions respectively.
- 3. Optimum injection pressure is 1.2 bar at 10 mm orifice diameter for CO, HC and NO_x emissions under load transient.
- 4. In case of load transient also, air injection doesn't increase the concentration of the exhaust emissions. 0.8%, 0.01% and 0.4% reduction is noted in the concentration of CO, HC and NO_x emissions respectively.
- 5. In order to achieve overall optimization of the system using air injection, it is found that 1 bar air injection at 5 mm orifice diameter and 1.2 bar air injection at 10 mm are optimum for speed and load transient respectively.

6. EFFECT OF AIR INJECTION ON EMISSION CHARACTERISTICS OF A CI ENGINE DURING TRANSIENT OPERATION

7

CONCLUSIONS

7.1 Major conclusions and thesis contribution

Main conclusions are drawn from the previous chapters and the contributions of the thesis are presented in the following section. The work presented in this thesis focused on the techniques to improve the transient response of turbocharged diesel engine. The techniques were employed through modelling and their impact on the engine performance was investigated. Comparative analysis was performed to observe the relative improvement in transient response brought by these techniques. Based on the analysis, air injection technique was selected and its effect was further investigated for various transient operating conditions and exhaust emissions.

The results obtained in this research are the outcomes of extensive numerical investigations, validated through experiment. Detailed diagrams were shown for speed drop and recovery period of the engine's transient response with respect to different operating parameter. This presents quantification of weightage and effect of each term.

7.1.1 Development of a novel strategy for turbo lag reduction

The transient response of a turbocharged diesel engine was investigated using three main methods: torque assist, reduction in inertia of compressor and air injection. The effect of these methods were investigated over a wide range of steady state and transient operating conditions to quantify interaction of turbocharger with engines through turbo lag phenomena. Fueling rate change representing the transient event was chosen for modelling the effect of these three methods. No such strategy was taken into consideration in the literature for the reduction of turbo lag.

Implementation of torque assistance by simulation had been scarcely reported in the

7. CONCLUSIONS

literature. Even though some experimental studies were previously conducted on torque assistance, Lotus Engine Simulation (LES) is used for the first time to investigate its impact on the transient performance. The investigations were performed to reduce the turbo lag using torque assistance and 0.16 Nm torque assistance was found to reduce the turbo lag considerably, thereby improving the transient response of turbocharged diesel engine. Another method of inertia reduction was also simulated using LES which benchmarks the result of the previous study. Marginal improvement was observed in the transient response of diesel engine when -10% inertia reduction is simulated. Finally air injection was simulated for fueling rate change event and a comparison was made between the results of the three methods. Air injection was found to bring maximum improvement in turbo lag reduction.

Due to the immense importance of diesel engine simulation modelling, the simulation based study using two advanced engine simulation software (Lotus Engine Simulation and Ricardo Wave) and the comparison between their results was one of the significant contribution of the present study.

7.1.2 Effect of air injection on the transient response of diesel engine under various operating conditions

Additional air injection at intake manifold significantly reduces the response interval of the turbocharger. This technology has paved way to carry out detailed investigations for a variety of test conditions. Effect of air injection on the response of heavy duty vehicles was scarcely reported. A wide range of operating conditions was considered to investigate the effect of air injection on the response of heavy duty turbocharged diesel engine. The parametric study had been undertaken to observe the response characteristics of the turbocharger under acceleration and load tests.

For acceleration tests, injection pressure was varied and the effect of varying injection pressure was observed on engine performance factors such as brake torque and brake power. These parameters were enhanced with injection pressure up to 2.5 bar. However no significant improvement was noticed after further increasing injection pressure (beyond 2.5 bar). Various tests were formulated to cover different operating conditions under acceleration tests. Optimum injection pressure was found depending upon the energy imparted with air injection. Turbo lag reduction/energy was calculated to find optimum value of air injection pressure for acceleration tests. 1 bar air injection was the optimum injection pressure with turbo lag reduction per unit energy as 0.290 per joule for compressor exit pressure, 0.392 per joule for turbine inlet pressure and 0.555 per joule for turbine inlet temperature (TIT). Quantitative achievement of turbo lag reduction per unit energy in this novel direction was itself a healthy contribution in the area of transient response improvement in turbocharged diesel engine.

For load tests, three different studies were conducted. The performance of the engine was investigated for constant load, load magnitude and load schedule. In the first study, the load was kept constant and a series of engine speeds were chosen based on the engine's operating range. Turbo lag reduction was observed for all the chosen speed range. Air injection brings maximum improvement in turbo lag reduction at lower speed range (1000-1400 rpm) while its effect on maximum attainable value was observed at higher speed range (1600-1800 rpm).

For the second and third study on load tests in Chapter 5, optimization of air injection was done based on the orifice diameter to regulate the mass flow of the air passing through the orifice. In case of load magnitude variation, air injection brings more improvement in lighter load application as compared to stronger load application. 50-70% load magnitude shows maximum improvement by air injection. Further, 3 bar air injection at 15 mm orifice diameter was more effective for 1 sec load schedule than for 2 sec. The results reveal that air injection into the intake manifold improves the turbo-lag under transient condition of load change.

7.1.3 Effect of air injection on environmental performance of the engine

A further noticeable effect of air injection technique was observed on the emission characteristics of the engine. Low speed transient - 1000 to 1700 rpm was taken into consideration and optimization of air injection was done to find the optimum value of air injection. The results of the simulation were evaluated for Euro 6 standard and 3 bar air injection at 10 mm orifice diameter was found to be the optimum injection pressure which brings slight reduction in CO and HC emissions and significant reduction in NO_x emissions. To evaluate the emissions for load tests, 50-70% load step in 1 sec was chosen and optimization was done in the same way as done for speed transient. Emissions were found to be below Euro 6 standard for all the injection pressures and orifice diameter. Quantification of emissions were done and 1.2 bar injection pressure at 10 mm orifice diameter was found to reduce all the three emissions (CO, HC and NO_x) as compared to no air injection.

It can be concluded from the study that although optimum injection pressure was different for different transient operating conditions, overall optimization was necessary. For acceleration tests, 1 bar air injection brings maximum improvement where as for the load tests 3 bar air injection at 15 mm orifice diameter was optimum. Looking

7. CONCLUSIONS

at exhaust emissions while applying air injection technique, CO and HC emissions satisfies Euro 6 standard under speed transient with 3 bar air injection at 10 mm orifice diameter while 1.2 bar air injection at 10 mm orifice diameter was the optimum value of air injection for reduced emissions under load transients. For overall optimization of the system, energy imparted per second is calculated for performance and emission characteristics under the transient operating conditions of speed and load changes. Based on this, 1 bar air injection at 5 mm orifice diameter and 1.2 bar air injection at 10 mm are optimum for speed and load transient respectively.

The analysis of the effect of air injection at intake manifold on exhaust emissions clearly indicate that air injection while improving the transient response of turbocharged diesel engine, does not negatively affects the engine emissions in the case of speed or load transients. This indicates that the research has been undertaken in the appropriate direction.

7.2 Limitations and scope for future work

The impact of the technologies discussed in the thesis has been focused mainly on computational basis. The study can provide useful data for designing the real system. It would be worth considering the design of experimental set up based on the outcomes of this research. A dedicated test rig can be designed for the implementation of this technique based on the control strategies proposed in the simulation platform.

Furthermore, since the focus of the thesis was on improving the transient response of diesel engine, hence emissions during transients were observed only to make sure that air injection technique does not need compromise on emissions. Further studies can be performed to evaluate the effect of air injection by simulating a particular driving cycle to meet emission standards. This would allow cost estimation and hence feasibility to implement this technology.

References

- Colin R Ferguson and Allan T Kirkpatrick. Internal combustion engines: applied thermosciences. John Wiley & Sons, 2015. 19, 21, 27
- [2] Safak Yildizhan, Vedat Karaman, Mustafa Ozcanli, and Hasan Serin. Calculation and optimizing of brake thermal efficiency of diesel engines based on theoretical diesel cycle parameters. *harvested by the following service*, page 100, 2016. 19
- [3] H Baderman. Boosting methods in highly downsized diesel engines. https://www.academia.edu/8241855/BOOSTING_METHODS_IN_HIGHLY_ DOWNSIZED_DIESEL_ENGINES, 2013. Accessed: 17.9.2019. 20
- [4] Vedat Karaman, Safak Yildizhan, Mustafa Ozcanli, and Hasan Serin. Calculation and optimizing of brake thermal efficiency of diesel engines based on theoretical diesel cycle parameters. *International Journal of Engineering Technologies*, 2(3): 100–104, 2016. 20
- [5] Constantine Rakopoulos, Athanasios Dimaratos, Evangelos Giakoumis, and Mark Peckham. Experimental assessment of turbocharged diesel engine transient emissions during acceleration, load change and starting. Technical report, SAE Technical Paper, 2010. 20, 24
- [6] Guangyong Zheng, Xiumin Yu, Jiangwei Liu, Tao Song, and Wenxiang Li. Performance development target setting of passenger car diesel engine. *Chinese Journal* of Mechanical Engineering, 26(3):512–517, 2013. 20
- [7] José Galindo, Hector Climent, Olivier Varnier, and Chaitanya Patil. Effect of boosting system architecture and thermomechanical limits on diesel engine performance: Part-i—steady-state operation. *International Journal of Engine Re*search, 19(8):854–872, 2018. 21

REFERENCES

- [8] Belachew Tesfa, Rakesh Mishra, C Zhang, F Gu, and AD Ball. Combustion and performance characteristics of CI (compression ignition) engine running with biodiesel. *Energy*, 51:101–115, 2013. 21, 27
- [9] Belachew Tesfa, Fengshou Gu, Rakesh Mishra, and AD Ball. Lhv predication models and lhv effect on the performance of CI engine running with biodiesel blends. *Energy conversion and management*, 71:217–226, 2013. 21
- [10] Dezong Zhao, Edward Winward, Zhijia Yang, Richard Stobart, and Thomas Steffen. Characterisation, control, and energy management of electrified turbocharged diesel engines. *Energy conversion and management*, 135:416–433, 2017. 21
- [11] Belachew Tesfa, Rakesh Mishra, Fengshou Gu, and AD Ball. Water injection effects on the performance and emission characteristics of a CI engine operating with biodiesel. *Renewable Energy*, 37(1):333–344, 2012. 21
- [12] Alex MKP Taylor. Science review of internal combustion engines. *Energy Policy*, 36(12):4657–4667, 2008. 21
- [13] Osama Hasib Ghazal. Air pollution reduction and environment protection using methane fuel for turbocharged CI engines. *Journal of Ecological Engineering*, 19 (5), 2018. 21
- [14] José Galindo, Hector Climent, Olivier Varnier, and Chaitanya Patil. Effect of boosting system architecture and thermomechanical limits on diesel engine performance: Part ii—transient operation. *International Journal of Engine Research*, 19(8):873–885, 2018. 21
- [15] Zhilong Hu, Kangyao Deng, Yi Cui, Xinxin Yang, and Baochuan Zhang. Steadystate and transient control strategies for a two-stage turbocharged diesel engine. *Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering*, 232(9):1167–1179, 2018. 21
- [16] Evangelos G Giakoumis. Review of some methods for improving transient response in automotive diesel engines through various turbocharging configurations. *Frontiers in Mechanical Engineering*, 2:4, 2016. 14, 21, 24, 25, 27, 28, 34, 35, 36, 37, 38, 40, 42, 46, 58, 119
- [17] Emmanuel P Kasseris and John B Heywood. Comparative analysis of automotive powertrain choices for the next 25 years. SAE Transactions, pages 626–647, 2007.
 21

- [18] John B Heywood and Orian Z Welling. Trends in performance characteristics of modern automobile SI and diesel engines. SAE International Journal of Engines, 2(1):1650–1662, 2009. 21
- [19] YangJun Zhang, Tao Chen, WeiLin Zhuge, ShuYong Zhang, and JianZhong Xu. An integrated turbocharger design approach to improve engine performance. *Sciences in China Series E: Technological Sciences*, 53(1):69–74, 2010. 21
- [20] A Kusztelan, D Marchant, Y Yao, Y Wang, S Selcuk, and A Gaikwad. Increases in low speed response of an IC engine using a twin-entry turbocharger. In Proceedings of the World Congress on Engineering, volume 3, pages 4–6, 2012. 21
- [21] Amir Reza Mahmoudi, Iman Khazaee, and Mohsen Ghazikhani. Simulating the effects of turbocharging on the emission levels of a gasoline engine. *Alexandria Engineering Journal*, 56(4):737–748, 2017. 14, 21, 22
- [22] Qiang Liu, Zhongchang Liu, Yongqiang Han, Jing Tian, Jun Wang, and Jian Fang. Experimental investigation of the loading strategy of an automotive diesel engine under transient operation conditions. *Energies*, 11(5):1293, 2018. 22, 32
- [23] LongPing Zhang, ZhongChang Liu, Jing Tian, and ShiJie Sun. Investigation of the combustion deterioration of an automotive diesel engine under transient operation. *Science China Technological Sciences*, 57(3):480–488, 2014. 22
- [24] Turbocharged diesel engine. Engine manual HudTier 37485 kW. 14, 23
- [25] J Shekaina and T Jayasingh. Novel turbocharger concept and control strategy for diesel passenger car. International Journal of Computer Applications, 41(2), 2012. 23
- [26] CO Katsanos, DT Hountalas, and TC Zannis. Simulation of a heavy-duty diesel engine with electrical turbocompounding system using operating charts for turbocharger components and power turbine. *Energy Conversion and Management*, 76:712–724, 2013. 23
- [27] Seungju Baek, Seungchul Woo, Youngkun Kim, and Kihyung Lee. Prediction of turbocharged diesel engine performance equipped with an electric supercharger using 1D simulation. *Energy*, 185:213–228, 2019. 25
- [28] ZC Liu, ZS Wang, J Li, and WX Li. Investigation on nox emission behavior from CA6DE1-21K diesel engine under transient operating conditions. *Journal* of Combustion Science and Technology, 10(3):193–196, 2006. 26

REFERENCES

- [29] ZC Liu, ZS Wang, J Li, and YH Wang. Smoke emission behavior of CA6DE1-21K diesel engine under transient operating conditions. *Journal of Combustion Science and Technology*, 10(6):484–488, 2004. 26
- [30] Lingge Sui, Zhongchang Liu, Yongqiang Han, and Jing Tian. Transient emission simulation and optimization of turbocharged diesel engine. *Frontiers in energy*, 7(2):237–244, 2013. 26
- [31] JR Serrano, FJ Arnau, V Dolz, and P Piqueras. Methodology for characterisation and simulation of turbocharged diesel engines combustion during transient operation. part 1: data acquisition and post-processing. Applied Thermal Engineering, 29(1):142–149, 2009. 26
- [32] JR Serrano, H Climent, C Guardiola, and P Piqueras. Methodology for characterisation and simulation of turbocharged diesel engines combustion during transient operation. part 2: phenomenological combustion simulation. Applied Thermal Engineering, 29(1):150–158, 2009. 26
- [33] Mohammad Hossein Vafaie, Behzad Mirzaeian Dehkordi, Payman Moallem, and Arash Kiyoumarsi. A new predictive direct torque control method for improving both steady-state and transient-state operations of the PMSM. *IEEE Transactions on power electronics*, 31(5):3738–3753, 2015. 26
- [34] Imad A Khalek, Matt Spears, and William Charmley. Particle size distribution from a heavy-duty diesel engine: Steady-state and transient emission measurement using two dilution systems and two fuels. SAE transactions, pages 137–147, 2003. 26
- [35] Ali Zare, Timothy A Bodisco, Md Nurun Nabi, Farhad M Hossain, MM Rahman, Zoran D Ristovski, and Richard J Brown. The influence of oxygenated fuels on transient and steady-state engine emissions. *Energy*, 121:841–853, 2017. 27
- [36] Zhongchang Liu, Xing Yuan, Jing Tian, Yongqiang Han, Runzhao Li, and Guanlong Gao. Investigation of sectional-stage loading strategies on a two-stage turbocharged heavy-duty diesel engine under transient operation with egr. *Energies*, 11(1):69, 2018. 27, 62
- [37] Anders Gusevik Gravdal. Diesel engine response improvements using hybrid turbocharging. Master's thesis, NTNU, 2017. 27

- [38] Jagadishsingh Bayas, Anita Wankar, and NP Jadhav. A review paper on effect of intake manifold geometry on performance of IC engine. International Journal of Advance Research and Innovative Ideas in Education, 2(2):101–106, 2016. 27
- [39] Getting started using lotus engine simulation. https: //lotusproactive.files.wordpress.com/2013/08/ getting-started-with-lotus-engine-simulation.pdf. Accessed: 16.10.2019. 28, 52, 63, 73, 76
- [40] CD Rakopoulos, AM Dimaratos, EG Giakoumis, and DC Rakopoulos. Evaluation of the effect of engine, load and turbocharger parameters on transient emissions of diesel engine. *Energy Conversion and Management*, 50(9):2381–2393, 2009. 14, 28, 44, 47
- [41] Shuo Liu, Yi Cui, Yi Wang, Kangyao Deng, and Sheng Liu. An evaluation method for transient response performance of turbocharged diesel engines. *Energy*, 2019. 31
- [42] EG Giakoumis and SC Lioutas. Diesel-engined vehicle nitric oxide and soot emissions during the european light-duty driving cycle using a transient mapping approach. Transportation Research Part D: Transport and Environment, 15(3): 134–143, 2010. 31
- [43] Constantine D Rakopoulos and EG Giakoumis. Review of thermodynamic diesel engine simulations under transient operating conditions. Technical report, SAE Technical Paper, 2006. 31
- [44] CD Rakopoulos, EG Giakoumis, and DT Hountalas. Experimental and simulation analysis of the transient operation of a turbocharged multi-cylinder IDI diesel engine. *International journal of energy research*, 22(4):317–331, 1998. 31
- [45] Kevin Koosup Yum, Nicolas Lefebvre, and Eilif Pedersen. An experimental investigation of the effects of cyclic transient loads on a turbocharged diesel engine. *Applied energy*, 185:472–481, 2017. 32, 45
- [46] S. He and Y. Meng. Simulation of automobile diesel engine under transient operating conditions. In Proceedings of 2011 International Conference on Electronic Mechanical Engineering and Information Technology, volume 7, pages 3571–3574, 2011. 32
REFERENCES

- [47] CD Rakopoulos and EG Giakoumis. Simulation and analysis of a naturally aspirated IDI diesel engine under transient conditions comprising the effect of various dynamic and thermodynamic parameters. *Energy Conversion and Management*, 39(5-6):465–484, 1998. 14, 32, 33
- [48] Fang Wei. A study on the improvement of marine diesel engine transient performance by means of air injection. Thesis of Hong Kong University, 2005. 34, 41, 46, 100
- [49] JD Ledger, Rowland S Benson, and H Furukawa. Improvement in transient performance of a turbocharged diesel engine by air injection into the compressor. SAE Transactions, pages 2412–2423, 1973. 34
- [50] Dariusz Cieslar. Control for transient response of turbocharged engines. PhD thesis, University of Cambridge, 2013. 34, 39, 40, 41, 47, 58
- [51] Chang Sik Lee and Nag Jung Choi. Effect of air injection on the characteristics of transient response in a turbocharged diesel engine. *International Journal of Thermal Sciences*, 41(1):63–71, 2002. 34, 41, 42
- [52] Evangelos G Giakoumis and Vasileios Tziolas. Modeling a variable-geometry turbocharged diesel engine under steady-state and transient conditions. *Journal* of Energy Engineering, 144(3):04018017, 2018. 37, 42
- [53] CD Rakopoulos, AM Dimaratos, and EG Giakoumis. Investigation of turbocharged diesel engine operation, exhaust emissions, and combustion noise radiation during starting under cold, warm, and hot conditions. Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering, 225(9):1118–1133, 2011. 37
- [54] Changryul Choi, Sunhyuk Kwon, and Sunghwan Cho. Development of fuel consumption of passenger diesel engine with 2 stage turbocharger. Technical report, SAE Technical paper, 2006. 37
- [55] Zizhong Ren, Thomas Campbell, and J Yang. Investigation on a computer controlled sequential turbocharging system for medium speed diesel engines. SAE transactions, pages 1842–1851, 1998. 37
- [56] Qingning Zhang, Andrew Pennycott, and Chris J Brace. A review of parallel and series turbocharging for the diesel engine. *Proceedings of the Institution*

of Mechanical Engineers, Part D: Journal of Automobile Engineering, 227(12): 1723–1733, 2013. 37

- [57] E Winward, J Rutledge, J Carter, A Costall, R Stobart, D Zhao, and Z Yang. Performance testing of an electrically assisted turbocharger on a heavy duty diesel engine. In *IMechE 12th International Conference on Turbochargers and Turbocharging*, pages 17–18, 2016. 37, 42
- [58] Nicola Terdich. Impact of electrically assisted turbocharging on the transient response of an off-highway diesel engine. PhD thesis, Imperial College London, 2014. 38
- [59] Gregor Tavčar, Frančišek Bizjan, and Tomaž Katrašnik. Methods for improving transient response of diesel engines-influences of different electrically assisted turbocharging topologies. Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering, 225(9):1167–1185, 2011. 39, 51
- [60] Diesel engine. http://www.academia.edu/8241855/BOOSTING_METHODS_IN_ HIGHLY_DOWNSIZED_DIESEL_ENGINES. Accessed: 16.1.2018. 39
- [61] Seiichi Ibaraki, Yukio Yamashita, Kunio Sumida, Hiroshi Ogita, and Yasuaki Jinnai. Development of the 'hybrid turbo,'an electrically assisted turbocharger. *Mitsubishi Heavy Ind. Tech. Rev*, 43(3):1–5, 2006. 39, 40, 76
- [62] N Watson. Transient performance simulation and analysis of turbocharged diesel engines. SAE Transactions, pages 1391–1409, 1981. 40
- [63] Oliver S Gilkes and Rakesh Mishra. Comparison of passive and active methods for improving transient performance of turbocharged engine systems. In Proceedings of Computing and Engineering Annual Researchers' Conference 2006: CEARC'06. University of Huddersfield, 2006. 40, 42, 46, 58, 82
- [64] Oliver S Gilkes, Rakesh Mishra, John D Fieldhouse, and HV Rao. Transient response of turbocharged diesel engine for transient operation using air injection assist system. In Proceedings of Computing and Engineering Annual Researchers' Conference 2007: CEARC'07. 2008. 41, 56
- [65] Shekaina and T Jayasingh. Novel configuration for air flow rationalization and turbo lag reduction in CRDI engine. International Journal of Scientific and Research Publications, 2(9):1–4, 2012. 42

REFERENCES

- [66] İbrahim Aslan Reşitoğlu, Kemal Altinişik, and Ali Keskin. The pollutant emissions from diesel-engine vehicles and exhaust aftertreatment systems. *Clean Technologies and Environmental Policy*, 17(1):15–27, 2015. 43, 123, 126
- [67] Belachew Tesfa, Fengshou Gu, Rakesh Mishra, and Andrew Ball. Emission characteristics of a CI engine running with a range of biodiesel feedstocks. *Energies*, 7(1):334–350, 2014. 43, 44, 55
- [68] Sui Ling-ge, Liu Zhong-chang, Han Yong-qiang, Shen Zhao-jie, and Zheng Guangyong. Discussion of transient response optimization strategies of turbocharged diesel engine under EGR step change operation. In 2011 International Conference on Computer Distributed Control and Intelligent Environmental Monitoring, pages 423–427. IEEE, 2011. 44
- [69] Qiang Liu, Jun Wang, and Yuanfu Ding. Optimization of fuel injection control strategy for a turbocharged diesel engine under transient condition. In 2018 7th International Conference on Energy, Environment and Sustainable Development (ICEESD 2018). Atlantis Press, 2018. 44
- [70] XG Wang, B Zheng, ZH Huang, N Zhang, YJ Zhang, and EJ Hu. Performance and emissions of a turbocharged, high-pressure common rail diesel engine operating on biodiesel/diesel blends. Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering, 225(1):127–139, 2011. 44
- [71] Masoud Aliramezani, CRKA Norouzi, and RE Hayes. A control oriented diesel engine NOx emission model for on board diagnostics and engine control with sensor feedback. *Proceedings of Combustion Institute-Canadian Section (CICS)*, 2019. 44
- [72] Ing David Blanco-Rodriguez. Modelling and observation of exhaust gas concentrations for diesel engine control. Springer, 2014. 45
- [73] Z Bazari. Diesel exhaust emissions prediction under transient operating conditions. SAE transactions, pages 1004–1019, 1994. 45, 47
- [74] N Kousheshi, S Jafarmadar, S Khalilarya, and M Bakhshi. Experimental studying the effect of air-injection on the combustion by an air-cell and emissions charactrestics in a di engine. http://www.ici.org.ir/FileForDownload/files/ Conference/4/papers/ICI-04-088.pdf, 2012. Accessed: 18.9.2019. 45

- [75] M Mansoury and S Jafarmadar. Experimental investigation of the effects of air injection and injection timing in a natural aspirated DI diesel engine. *International Journal of Engineering*, pages 1025–2495, 2013. 45
- [76] Xiaobo Zhou, Erxi Liu, Dezeng Sun, and Wanhua Su. Study on transient emission spikes reduction of a heavy-duty diesel engine equipped with a variable intake valve closing timing mechanism and a two-stage turbocharger. *International Journal of Engine Research*, 20(3):277–291, 2019. 45
- [77] Shuzhan Bai, Jianlei Han, Min Liu, Shunshun Qin, Guihua Wang, and Guo-xiang Li. Experimental investigation of exhaust thermal management on NOx emissions of heavy-duty diesel engine under the world harmonized transient cycle (WHTC). *Applied Thermal Engineering*, 142:421–432, 2018. 45
- [78] Wei Guan, Vinícius Pedrozo, Hua Zhao, Zhibo Ban, and Tiejian Lin. Investigation of EGR and miller cycle for NOx emissions and exhaust temperature control of a heavy-duty diesel engine. Technical report, SAE Technical Paper, 2017. 45
- [79] Timothy V Johnson. Review of diesel emissions and control. SAE International Journal of Fuels and Lubricants, 3(1):16–29, 2010. 45
- [80] Vicente Bermúdez, José Ramón Serrano, Pedro Piqueras, and Óscar García-Afonso. Analysis of heavy-duty turbocharged diesel engine response under cold transient operation with a pre-turbo aftertreatment exhaust manifold configuration. International Journal of Engine Research, 14(4):341–353, 2013. 45
- [81] Rakesh Mishra and Syed Mohammad Saad. Simulation based study on improving the transient response quality of turbocharged diesel engines. *Journal of Quality* in Maintenance Engineering, 23(3):297–309, 2017. 47, 52
- [82] Constantine D Rakopoulos, Athanasios M Dimaratos, EG Giakoumis, and Dimitrios C Rakopoulos. Exhaust emissions estimation during transient turbocharged diesel engine operation using a two-zone combustion model. *International Journal* of Vehicle Design, 49(1-3):125–149, 2009. 47
- [83] Y Cui, K Deng, and J Wu. A modelling and experimental study of transient NOx emissions in turbocharged direct injection diesel engines. Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering, 218(5):535-541, 2004. 47

REFERENCES

- [84] J Benajes, JM Lujan, V Bermudez, and JR Serrano. Modelling of turbocharged diesel engines in transient operation. part 1: insight into the relevant physical phenomena. Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering, 216(5):431-441, 2002. 49
- [85] CD Rakopoulos and EG Giakoumis. Availability analysis of a turbocharged diesel engine operating under transient load conditions. *Energy*, 29(8):1085–1104, 2004.
 49
- [86] Nikolaos F Sakellaridis, Spyridon I Raptotasios, Antonis K Antonopoulos, Georgios C Mavropoulos, and Dimitrios T Hountalas. Development and validation of a new turbocharger simulation methodology for marine two stroke diesel engine modelling and diagnostic applications. *Energy*, 91:952–966, 2015. 49
- [87] KinYip Chan, Andrzej Ordys, Konstantin Volkov, and Olga Duran. Comparison of engine simulation software for development of control system. *Modelling and Simulation in Engineering*, 2013:5, 2013. 49
- [88] RJ Pearson, MD Bassett, NP Fleming, and T Rodemann. Lotus engineering software–an approach to model-based design. In *The 2002 North American ADAMS Conference in Scottsdale, Arizona*. Citeseer, 2002. 51, 52
- [89] JB Heywood. 1988, Internal combustion engine fundamentals, mcgraw-hill, new york. 51, 55, 56
- [90] Hasan Üstün Başaran. A simulation based study to improve active diesel particulate filter regeneration through waste-gate valve opening modulation. International Journal of Automotive Science and Technology, 3(2):32–41, 2019. 52
- [91] Thermodynamics, Fluid Mechanics Group, and WJD Annand. Heat transfer in the cylinders of reciprocating internal combustion engines. *Proceedings of the Institution of Mechanical Engineers*, 177(1):973–996, 1963. 52
- [92] Diesel combustion. Ricardo WAVE knowledge center help files. 54, 55, 57, 58, 126
- [93] Belachew Tesfa, Rakesh Mishra, Feng Shou Gu, and Andrew Ball. Condition monitoring of CI engine running on biodiesel using transient process. In Key Engineering Materials, volume 413, pages 495–503. Trans Tech Publ, 2009. 57
- [94] Martin Turesson. Modelling and simulation of a two-stroke engine. Master's thesis, 2009. 58

- [95] Wave simulink co-simulation. https://www.mathworks.com/products/ connections/product_detail/wave.html. Accessed: 17.9.2019. 58
- [96] Isaac H Brito. Modeling and control of wastegate equipped turbocharged engines. Master's thesis, 2010. 58
- [97] Nag-Jung Choi and Seong-Mo Oh. An experimental study on the improvement of turbocharger lag by means of air injection in a turbocharged diesel engine. *Journal of the Korean Society of Marine Engineering*, 34(7):951–962, 2010. 59, 63
- [98] Belachew Tesfa, Rakesh Mishra, Fengshou Gu, and Oliver Gilkes. Emission behavior of a CI engine running by biodiesel under transient conditions. Technical report, SAE Technical Paper, 2010. 14, 61
- [99] Constantine D Rakopoulos and Evangelos G Giakoumis. Diesel engine transient operation: principles of operation and simulation analysis. Springer Science & Business Media, 2009. 65, 113
- [100] Belachew Chekene Tesfa. Investigations into the performance and emission characteristics of a biodiesel fuelled CI engine under steady and transient operating conditions. PhD thesis, University of Huddersfield, 2011. 69
- [101] R. B. Abernethy, R. P. Benedict, and R. B. Dowdell. ASME Measurement Uncertainty. *Journal of Fluids Engineering*, 107(2):161–164, 06 1985. ISSN 0098-2202. doi: 10.1115/1.3242450. URL https://doi.org/10.1115/1.3242450. 69
- [102] Measurements and error analysis. https://www.webassign.net/question_ assets/unccolphysmechl1/measurements/manual.html. Accessed: 6.11.2019. 69
- [103] ZC Liu, KB Yu, J Tian, YQ Han, SL Qi, and PK Teng. Influence of rail pressure on a two-stage turbocharged heavy-duty diesel engine under transient operation. *International Journal of Automotive Technology*, 18(1):19–29, 2017. 76
- [104] SM Saad and Rakesh Mishra. Performance of a heavy-duty turbocharged diesel engine under the effect of air injection at intake manifold during transient operations. Arabian Journal for Science and Engineering, 44(6):5863–5875, 2019. 86
- [105] Walter Knecht. Diesel engine development in view of reduced emission standards. Energy, 33(2):264–271, 2008. 123